

No. 1988 April 2024

The economic dynamics of city structure: Evidence from Hiroshima's recovery

Kohei Takeda Atsushi Yamagishi



THE LONDON SCHOOL OF ECONOMICS AND POLITICAL SCIENCE



Economic and Social Research Council

Abstract

We provide new theory and evidence on the resilience of internal city structure after a large shock, analyzing the atomic bombing of Hiroshima. Exploiting newly digitized data, we document that the city structure recovered within five years after the bombing. Our new dynamic quantitative model of internal city structure incorporates commuting, forward-looking location choices, migration frictions, agglomeration forces, and heterogeneous location fundamentals. Strong agglomeration forces in our estimated model explain Hiroshima's recovery, and we find an alternative equilibrium where the city center did not recover. These results highlight the role of agglomeration forces, multiple equilibria, and expectations in urban dynamics.

Keywords: agglomeration, history, expectations, atomic bombing, spatial dynamics JEL Codes: C73; N45; O18; R12; R23

This paper was produced as part of the Centre's Urban Programme. The Centre for Economic Performance is financed by the Economic and Social Research Council.

We are grateful to Steve Redding for his continued guidance and support. We particularly thank Leah Boustan and Nobuhiro Kiyotaki for their numerous helpful comments, along with Elena Aguilar, Gabriel Ahlfeldt, Treb Allen, Mark Bamba, Eduard Boehm, Jaedo Choi, Levi Crews, Ellora Derenoncourt, Klaus Desmet, Farid Farrokhi, Mayara Felix, Allison Green, Gene Grossman, Richard Hornbeck, Allan Hsiao, Hiroyuki Kasahara, Guy Michaels, Yuhei Miyauchi, Eduardo Morales, Tomoya Mori, David Nagy, Kentaro Nakajima, Georgios Nikolakoudis, Ezra Oberfield, Gianmarco Ottaviano, Pablo Ottonello, Luigi Pascali, Sebastian Roelsgaard, Esteban Rossi-Hansberg, Yasuhiro Sato, Daniel Sturm, Katsuya Takii, Kensuke Teshima, Lin Tian, Carolina Villegas-Sanchez, David Weinstein, Chansik Yoon, Xuanli Zhu, and seminar and conference participants for their feedback. We thank Chugoku Region Development Bureau and Ministry of Land, Infrastructure, Transport and Tourism (MLIT) for providing access to the person-trip survey data. We acknowledge financial support from the International Economics Section, Princeton University, the Institute of Economic Research, Hitotsubashi University, National University of Singapore, and the Funai Foundation for Information Technology. We also would like to thank Hiroki Chiba for excellent research assistance and Shinji Nakano and Takahiro Sato for their help with the GIS data. An earlier version of this paper was circulated under the title "History versus Expectations in the Spatial Economy: Lessons from Hiroshima." We take sole responsibility for errors in this paper.

Kohei Takeda, National University of Singapore and Centre for Economic Performance at London School of Economics. Atsushi Yamagishi, Princeton University.

Published by Centre for Economic Performance London School of Economic and Political Science Houghton Street London WC2A 2AE

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means without the prior permission in writing of the publisher nor be issued to the public or circulated in any form other than that in which it is published.

Requests for permission to reproduce any article or part of the Working Paper should be sent to the editor at the above address.

1 Introduction

Cities have faced a host of shocks throughout history. Wars, natural disasters, pandemics, and technological shocks have all impacted city structure – the spatial distribution of economic activities within cities. However, there remains substantial debate about whether city structure is resilient to large shocks and what mechanisms are behind this resilience (Glaeser 2022). Theoretically, resilience of city structure after temporary shocks emerges from exogenous locational fundamentals that uniquely determine the distribution of economic activities, or the presence of strong agglomeration forces by which the city structure is determined via a coordination of expectations around the focal point. However, the empirical importance of these different mechanisms remains an open question (Lin and Rauch 2022).

Understanding the mechanisms that underlie the resilience of city structure would aid the reconstruction of war-torn cities, improve urban revitalization efforts, and inform planning for future shocks. Nevertheless, there are two important challenges in answering these questions. First, we rarely observe a large shock to city structure that allows us to capture spatially-granular data on economic activities over a long period of time. Second, we need a quantifiable model of the dynamics of internal city structure that allows us to disentangle the mechanisms underlying the resilience of city structure.

In this paper, we provide new theory and evidence on the resilience of city structure by analyzing the atomic bombing of Hiroshima, one of the most remarkable examples of urban resilience in human history. The atomic bombing completely destroyed the city center while sparing its outskirts. This distinctive and massive shock provides a unique laboratory for studying the dynamics of city structure. We collect and digitize new granular historical data on the distribution of economic activities within Hiroshima. Using this data, we first document that the city structure of Hiroshima recovered to its pre-war state within five years after the bombing. We then construct and calibrate a new dynamic quantitative model of internal city structure, which explains this observed recovery. In our estimated model, strong agglomeration forces created by population and employment density yield better local amenities and increase productivity. This provided the key incentive for people to again live and work in the city center. Finally, we show that these agglomeration forces induce multiple equilibria. In particular, there exists an alternative equilibrium in which the city center did not recover, in contrast to the observed recovery equilibrium. We argue that self-fulfilling expectations of recovery might have played an important role in realizing the recovery equilibrium.

We begin by describing the historical context and our newly collected data on the distribution of economic activities within Hiroshima. As of 1945, most the administrative region of Hiroshima lay within 6 kilometers of the city center. On August 6, 1945, the atomic bomb hit near the city center and destroyed almost all structures within 2 kilometers of the city center, but many structures on the outskirts of the city were much less affected. Some areas on the outskirts even experienced an increase in population due to inflows of survivors from the city center. Consequently, the atomic bombing was an extremely large shock to city structure; the pre-war city center now had the lowest population and employment density in the city. To conduct our quantitative analysis at a spatially granular level, we collect and digitize new historical data on population, employment, wartime destruction, and fundamental characteristics at the city block level within Hiroshima. Importantly, our dataset covers both the pre-bombing and immediate post-bombing periods, allowing us to investigate the recovery of central Hiroshima in detail.

Through descriptive and reduced-form analyses, we reveal that the city structure of Hiroshima was resilient to this unprecedented shock. Our findings are twofold: (i) the destroyed city center again became the main hub of economic activity within five years of the atomic bombing; and (ii) the recovery of central Hiroshima is not explained by various observed fundamental location characteristics, which could directly affect amenities and productivity independently of the local density of economic activities (e.g., altitude, access to natural water). There are two possible explanations for these findings. First, while our results suggest that the recovery of the city center is not explained by its observed locational advantages and our results also hold even within a small homogeneous area, it is still possible that the destroyed city center retained some unobserved locational advantages that survived the bombing (e.g., scenic views). Second, people may have expected the recovery of the destroyed city center when making location choices, and the incentive to again live and work in the city center came from agglomeration forces due to expected high density as in the pre-war period. We analyze these two possible explanations using our structural model.

We develop a new dynamic quantitative model of internal city structure. Our model is the first quantitative urban model that accommodates commuting, forward-looking location choices, migration frictions, agglomeration forces, and heterogeneous location fundamentals. The commuting patterns within a city are endogenously determined by individual choices of workplace and residence. Individuals correctly anticipate the future path of the economy when making location decisions, and neighborhood amenities and productivity depend on population and employment density. Location-specific fundamental amenities and productivity capture the heterogeneous advantages of locations in a city that are independent of population and employment density. In addition, our model incorporates migration frictions, which induce history-dependence in city structure. These model elements are necessary to capture alternative possible determinants of the dynamics of internal city structure.

We calibrate our model and evaluate its ability to explain the recovery of central Hiroshima. We estimate the model using the observed location choices of Hiroshima residents from 1955 to 1975, after the period of rapid recovery, because sufficient data for calibrating our model are only available for this time period. We leverage the structure of the model to estimate model parameters and compute unobserved location characteristics. Our key parameters describe how the agglomeration forces increase amenities and productivity as population and employment density rise. We estimate these forces under the identification assumption that exogenous changes in the amenities and productivity of each block over time are uncorrelated with distance from the city center, while allowing for arbitrary heterogeneity in location-specific amenities and productivity. We find strong agglomeration forces in both amenities and productivity. We then assess how well our calibrated model fits the location choice data for 1950 when people had returned to the destroyed city center. The endogenous mechanisms of our model successfully account for the resurgence observed in the data for 1950.

We highlight that agglomeration forces play a key role in explaining the recovery of central Hiroshima. To this end, we simulate a counterfactual population and employment distribution in which agglomeration forces in amenities and productivity are absent and fundamental productivity and amenities for 1950 are equal to their estimated averages from the 1955–1975 data. We find that our calibrated model without agglomeration forces cannot predict the recovery of the city center, consistent with the idea that agglomeration forces played an important role in the resurgence of the center. Theoretically, in the presence of strong agglomeration forces, there is potential for multiple equilibria because the city center would not be attractive if its recovery were not expected in the near future. To investigate this, we numerically solve the model for an alternative rational expectations equilibrium in which the population and employment densities of the city center do not recover. This suggests that the observed pattern of recovery is one equilibrium selected from multiple, and that self-fulfilling expectations of recovery might be crucial in selecting this equilibrium. We argue that certain factors, such as government recovery plans, the anchoring effect of salient location characteristics in the city center (e.g., tram networks, the destroyed Hiroshima castle), property rights, and popular narratives of rebuilding, may have induced expectations that the destroyed city center would return to its high density as in the pre-war period. Our results suggest the importance of these factors in the rapid recovery of city structure through fostering the formation of such recovery expectations.

Overall, our analysis of Hiroshima highlights the role of agglomeration forces, multiple equilibria, and expectations in the economic dynamics of city structure. For cities recovering from a large shock, our findings indicate the importance of agglomeration forces and creating potentially self-fulfilling expectations for recovery. Our results further suggest that policymakers could substantially change the dynamics of city structure if they could influence expectations about a city's future.

This paper contributes to studies on the determinants of the concentration of economic ac-

tivity in space, which is at the core of urban economics and economic geography. A theoretical literature has uncovered the importance of fundamental location characteristics and agglomeration forces in shaping the spatial distribution of economic activities, including Fujita and Ogawa (1982), Fujita and Thisse (1996), Fujita, Krugman, and Venables (1999), Lucas and Rossi-Hansberg (2002) and Ahlfeldt, Redding, Sturm, and Wolf (2015). Moreover, when agglomeration forces are important relative to heterogeneity in location characteristics so that there are multiple equilibria, initial conditions ("history") or self-fulfilling expectations may determine the spatial distribution of economic activities by selecting a particular equilibrium. Krugman (1991) and Matsuyama (1991) show that self-fulfilling expectations can induce a transition from one steady state to another when multiple equilibria exist, implying that the initial conditions determined by history can be overcome by expectations.¹ We empirically contribute to this discussion by analyzing the atomic bombing of Hiroshima as a large exogenous shock to city structure and showing the importance of agglomeration forces, multiple equilibria, and expectations in the economic dynamics of city structure.²

Many empirical studies have investigated the importance of historical shocks as determinants of the spatial distribution of economic activities. Previous studies exploiting war-time destruction across cities and regions have typically found that large shocks have only temporary impacts, including Davis and Weinstein (2002, 2008), Brakman, Garretsen, and Schramm (2004), Bosker, Brakman, Garretsen, and Schramm (2007), Miguel and Roland (2011), Feigenbaum, Lee, and Mezzanotti (2022). Among others, Davis and Weinstein (2002) finds that the population distribution *across* Japanese cities after World War II converged with its pre-war trend, including Hiroshima. Yet, studies that exploit shocks other than war-time destruction often find that large shocks can have persistent or permanent effects on the spatial distribution of economic activities, including Redding, Sturm, and Wolf (2011), Bleakley and Lin (2012), Schumann (2014), Siodla (2015), Hornbeck and Keniston (2017), Michaels and Rauch (2018), Brooks and Lutz (2019), Ambrus, Field, and Gonzalez (2020), Heblich, Trew, and Zylberberg (2021), Allen and Donaldson (2022), Brooks, Rose, and Veuger (2022) and Yamagishi and Sato (2023).³ Our paper is distinctive from these studies in three important ways. First, we analyze the spatial distribution of economic activities *within* a

¹Among others, see Fukao and Bénabou (1993), Rauch (1993), Holmes (1999), Baldwin (2001), Ottaviano (2001), Oyama (2009), and Barreda-Tarrazona, Kundu, and Østbye (2021) for developments in self-fulfilling expectations and economic geography. Self-fulfilling expectations also matter in other important economic contexts with multiple equilibria, including structural transformation in economic development (Murphy, Shleifer, and Vishny 1989).

²Fujita and Thisse (1996) states "[a]nother reason for [a spatial structure's] inertia...is the formation of self-fulfilling prophecies about the development of some areas. Indeed, it seems reasonable to consider existing cities as focal points that help agents coordinate their spatial decisions. In such a context, reshaping the urban landscape would then require major changes in agents' expectations."

³Harada, Ito, and Smith (forthcoming) and Redding and Sturm (2023) estimate the long-run impact of bombing on neighborhood quality within Tokyo and London. Compared to them, we do not analyze neighborhood quality as such data is unavailable and instead focus on the dynamics of the re-emergence of city structure.

city using new spatially granular data. Second, we use the atomic bombing of Hiroshima as an exogenous and unprecedentedly large shock to the internal structure of a city. Third, and most importantly, we develop and apply a novel dynamic quantitative urban model to this historical shock to investigate why we observe such resilience in Hiroshima, highlighting the importance of agglomeration forces and self-fulfilling expectations in overcoming the catastrophe. In particular, their importance can reconcile the aforementioned empirical studies that are split on the persistence of historical shocks: we often observe history independence in the distribution of economic activities because expectations of recovery to the pre-event situation tend to emerge after wartime destruction. History dependence may arise in other contexts where such expectations are absent.

Our structural analysis also relates to recent advancements in quantitative spatial models, as reviewed in Redding and Rossi-Hansberg (2017). To analyze the resilience of city structure, we develop a new quantitative urban model with commuting, forward-looking location choices, migration frictions, agglomeration forces, and heterogeneous location fundamentals. Studies that consider commuting and agglomeration forces within cities (Ahlfeldt et al. 2015; Monte, Redding, and Rossi-Hansberg 2018; Dingel and Tintelnot 2020; Tsivanidis 2022) do not accommodate forward-looking migration decisions and migration frictions, while those with forward-looking migration decisions (Caliendo, Dvorkin, and Parro 2019; Balboni 2021; Heblich, Trew, and Zylberberg 2021; Warnes 2021; Allen and Donaldson 2022; Almagro and Domínguez-Iino 2022; Kleinman, Liu, and Redding 2023) do not incorporate commuting or agglomeration forces. Among others, Monte, Porcher, and Rossi-Hansberg (2023) presents a model for one-dimensional cities on the real line with forward-looking individuals who choose work arrangements and commuting patterns. In contrast, our model allows for an arbitrary number of discrete locations within cities and location fundamentals, which are tractable when mapping the data to the model.⁴ Importantly, we integrate commuting, forward-looking location choices, migration frictions, agglomeration forces, and heterogeneous location fundamentals into a single framework that is otherwise parsimonious since data availability in historical contexts is often limited. The tractability of the model is particularly useful in data-scarce environments.

Lastly, this paper relates to studies on the recovery of Hiroshima from the atomic bombing (Hiroshima City Government, 1971; 1983). There is little econometric analysis on the distribution of economic activities within the city and the resurgence of the city center. Our paper formally analyzes the recovery pattern using newly-digitized granular historical data and a novel quantitative economic model. This provides new evidence on the resilience of Hiroshima's city structure and a new approach to understanding the economic mechanisms behind the resilience.

⁴As an early theoretical contribution, Rossi-Hansberg (2004) presents the canonical urban model with shocks to internal city structure.

The rest of the paper is structured as follows. Section 2 describes the historical context and data. Section 3 presents the reduced-form analysis and Section 4 introduces the theoretical framework. Section 5 calibrates the model and demonstrates that our model accurately fits the recovery of central Hiroshima. In Section 6 we undertake a counterfactual analysis to show the roles of agglomeration forces and expectations in the recovery. Section 7 explores the potential factors contributing to the formation of expectations. Section 8 concludes.

2 Historical Background and Data

This section briefly describes the history of Hiroshima prior to the atomic bombing and the impact of the bombing on the city (Subsection 2.1) and how we construct new granular spatial data on population, employment and other characteristics of Hiroshima (Subsection 2.2).

2.1 Historical Background

The development of Hiroshima started in the late 16th century when Terumoto Mōri, a local samurai lord, built Hiroshima castle.⁵ Hiroshima has been a major city in the *Chugoku* region of Japan since then thanks to its proximity to the sea and rivers. Early in the 20th century, the city grew quickly. In 1935, 310,118 people lived in Hiroshima, which made it the seventh-largest city in Japan by population. As Japan gradually transitioned to a war economy through the Second Sino-Japanese War (1937–1945) and Pacific War (1941–1945), growth slowed and then reversed. Before the atomic bombing, the city had an estimated population of 350,000. As the U.S. overwhelmed Japan during World War II, most Japanese cities endured extensive non-atomic air raids (Davis and Weinstein 2002). However, the U.S. avoided bombing Hiroshima to preserve the city as the "best laboratory" for demonstrating the effects of the atomic bomb.⁶ Consequently, the atomic bombing was essentially the only direct destruction the city experienced during WWII.

On August 6, 1945, the U.S. Air Force dropped the atomic bomb "Little Boy" near the center of Hiroshima. The damage to people and buildings was unprecedentedly catastrophic. The city government of Hiroshima estimates that 140,000 people died by the end of 1945 as a result of the atomic bombing, although it is difficult to determine the exact number.⁷ The death rate was near

⁵In this paper we use the word "Hiroshima" to refer to the administrative Hiroshima City (*Hiroshima-shi*). We sometimes explicitly state Hiroshima City to clearly distinguish it from Hiroshima Prefecture.

⁶Based on "Minutes of the Second Meeting of the Target Committee Los Alamos, May 10-11, 1945" (http://www. dannen.com/decision/targets.html, last accessed on October 28, 2023), the target of the bombing was determined based on the city's size and its flat terrain to best measure the damage from the bombing. Notably, local economic conditions were not considered in selecting targets.

⁷The real death toll is likely to be even higher because the atomic bombing caused severe injuries and diseases that killed many after 1945. Source: https://www.city.hiroshima.lg.jp/site/english/9803.html (last accessed on October 28, 2023). This damage was much more severe than in other cities that endured extensive air raids. For example, the population of Tokyo was approximately 7 million in 1940. U.S. air raids on Tokyo killed over 100,000 civilians and

Figure 1: Destruction of the atomic bombing in Hiroshima



Note: Panel (a) is a photograph from the United States Strategic Bombing Survey made available by the U.S. National Archives and Records Administration. Panel (b) shows a map of Hiroshima at the time of the bombing, along with block-level data (197 blocks in total) on the fraction of completely destroyed buildings and the epicenter (Hiroshima City Government 1971; Takezaki and Soda 2001). Remote islands (*Nino-shima, Kanawa-jima, Touge-shima*) are omitted for better visibility. We use as the background image the 1950 topographic map taken from the Time Series Topographic Map Viewer of Japan (Tani 2017, https://ktgis.net/kjmapw/).

100 percent for those within 1 kilometer of the epicenter. The bomb also destroyed a large number of buildings: 51,787 out of 76,237 buildings in Hiroshima were completely destroyed and 18,720 were partly destroyed. The vast majority of buildings within 2 kilometers of the city center were completely destroyed. This can be seen in Figure 1a, which was taken near the epicenter of the bombing. The population of Hiroshima dropped to 136,518 as of November 1945, about one-third of the pre-war population.

In contrast to the total destruction in central Hiroshima, the outskirts of the city were much less damaged. Figure 1b shows the fraction of completely destroyed buildings at the block level (Hiroshima City Government 1971; Takezaki and Soda 2001). While nearly all buildings in the dark-colored areas close to the epicenter were destroyed, buildings in the light-colored areas away from the epicenter were much less damaged. As a result, the outskirts of Hiroshima experienced a significant increase in population as survivors from the city center flooded in. As of November 1, 1945, Hiroshima beyond 3 kilometers from the epicenter had 142 percent of its pre-bombing population.

The war ended on August 15, 1945. People initially doubted whether Hiroshima could recover.

damaged approximately 700,000 housing units. Source: https://tokyo-sensai.net/about/tokyoraids/ (In Japanese, last accessed on October 28, 2023). While the absolute numbers are large in Tokyo, the percentage rates in Hiroshima are substantially greater than in Tokyo due to its smaller population.

Despite then-limited scientific knowledge, radioactive contamination was a major concern immediately after the bombing. Rumors circulated that "nothing will grow here for 75 years."⁸ However, the serious radioactive contamination caused by the bombing decayed rapidly. According to the Hiroshima City government, radiation at the epicenter was 1/1,000th a day after the bombing and 1/1,000,000th a week later.⁹ Furthermore, a large typhoon hit Hiroshima on September 17, 1945, about six weeks after the bombing. According to the U.S. Atomic Bomb Casualty Commission, the typhoon probably washed away much contaminated material, bringing radioactivity down to a relatively safe level (Takahashi 2008). Given this evidence, we do not consider potential radioactive contamination in analyzing the recovery of Hiroshima because living in Hiroshima was unlikely to be meaningfully unhealthy after this typhoon.

Despite initial pessimism, people gradually became optimistic about Hiroshima's destroyed city center (Hiroshima City Government 1971). Although it is difficult to identify one single factor, several may have contributed to such optimism. First, the city government of Hiroshima published a recovery plan. Even though the government faced a serious budget shortage and could not implement most of its plan for several years after the bombing, the plan might have induced optimism about Hiroshima's future.¹⁰ Second, the presence of salient location characteristics in the city center, such as the transportation network and the destroyed Hiroshima castle, may have anchored people's expectations for recovery despite the severe damage. Third, pre-war private property ownership was preserved, though almost all homeowners near the city center lost their homes and many landowners and homeowners close to the city center were killed by the bombing. Finally, rebuilding narratives may have been shared by people and coordinated their expectations. We revisit the discussion about the formation of expectations in Section 7.

Notwithstanding the lack of strong public actions, Hiroshima experienced a strong recovery due to private efforts. In 1955, Hiroshima had a population of 357,287, larger than the 1935 population. Hiroshima continued to grow and physically expanded along the way.¹¹ Today, Hiroshima has a population of approximately 1.2 million, making it the 10th largest Japanese municipality and the largest in the *Chugoku* region of Japan.

¹¹As shown in Davis and Weinstein (2002), the aggregate city population recovered its pre-war trend around twenty years after the atomic bombing.

⁸Source: https://www.bbc.com/news/world-asia-53660059, last accessed on October 28, 2023.

⁹Source: https://www.city.hiroshima.lg.jp/site/english/9809.html, last accessed on October 28, 2023.

¹⁰The recovery council was formed in February 1945 and consisted of 26 members, including the former mayor, city councillors, and local business leaders. However, despite the exceptionally catastrophic damage, Hiroshima could not get special budgetary treatment until the enactment of the Hiroshima Peace Memorial City Construction Law in 1949. In 1947, the budget for the reconstruction of Hiroshima City was 0.56 billion JPY, which was only 2.5 percent of the estimated total budget of 23 billion JPY (Shinoda 2008). The city focused on providing public housing, but it could only provide 3,000 units during 1945–1950, relative to the over 70,000 buildings destroyed. The restoration of pre-war infrastructure was also prioritized, but this did provide a particular advantage to the city center as the city outskirts also had comparable infrastructure.

2.2 Data

We have collected and digitized various sources of information on economic activity in Hiroshima before and after the war. Here, we provide a brief overview of our data sources. Appendix A provides further details.

Spatial Units The spatial unit of our analysis is a city block (*cho-cho-moku*) in Hiroshima. As our primary definition of city blocks, we use the block boundaries as of the bombing constructed by Takezaki and Soda (2001). In comparing the pre-war and post-war periods, we focus on areas that were part of Hiroshima as of the bombing.¹² Throughout this paper, we use the block definitions of 1945. The pre-war central business district (CBD) is defined as the mid-point of the *Kamiya-cho* block and *Hacchobori* blocks, which were the two prominent central areas of pre-war Hiroshima. Note that the recovery of the city center documented below implies that the pre-war city center corresponds to the post-recovery city center. We address the revisions of the block boundaries over time by converting all data to the 1945 block definitions based on areal weighting interpolation, and we digitize the block boundaries of 1966 and 1976 to implement this. Throughout the paper, the number of blocks is 174 and the average size of blocks is 0.32 square kilometers.¹³

Destruction by the Atomic Bombing We primarily use the fraction of completely destroyed buildings as a measure of the severity of destruction. The block-level destruction rate is reported in Hiroshima City Government (1971). We augment the digitization of Takezaki and Soda (2001) by consulting Hiroshima City Government (1971) to correct typos in their data and obtain additional information on missing values. Panel (b) of Figure 1 in the previous section illustrates the share of completely destroyed buildings in each block.

Population We collect and digitize population data at the block level. We refer to the Statistical Handbook of Hiroshima (*Hiroshima-shi toukei sho*) for 1933–1936 and the Statistical Abstract of Hiroshima (*Hiroshima shisei youran*) for 1945–1953.¹⁴ From 1955, the population data is taken from the Population Census. Panel (a) of Figure 2 provides a visualization of Hiroshima's population over time. The population of Hiroshima was increasing prior to the atomic bombing, dropped significantly after the bombing, and resumed growth again after WWII. We also observe that the

¹²The city boundaries gradually expanded since 1955 through municipal mergers as the Hiroshima metropolitan area grew. The administrative Hiroshima as of 1945 roughly corresponds to the four central wards (*Naka-ku*, *Nishi-ku*, *Minami-ku*, *Higashi-ku*) of Hiroshima today. The expansion of the administrative boundaries and commuting zones of Hiroshima implies that our data is more concentrated within the relatively central locations as time elapses.

¹³A Japanese city block is generally smaller than a U.S. census tract, which has a population of around 4,000, but larger than a U.S. census block, which has 40 housing units. In our data on Hiroshima, the average area size for blocks is 0.04 (0.13) square kilometers within 1 (3) kilometer of the CBD, and the average block area is 2.19 square kilometers among blocks more than 3 kilometers from the CBD.

¹⁴For 1945–1950, population is reported using a more aggregated block definition. We combine this information with the block-level destruction rate of buildings to predict the block-level population distribution.

center of Hiroshima declined as a share of the population over time, reflecting the suburbanization of the city that absorbed most post-war population growth. Note that this declining trend was already observed pre-WWII, suggesting that a smaller central population share after WWII does not necessarily mean that recovery was incomplete.

Employment We collect and digitize employment data at the block level from various sources.¹⁵ For 1938, we refer to the Survey of Commerce and Industry in Hiroshima (*Hiroshima-shi shoukou-gyou keiei chousa*) which records the number of establishments at the block level. The number of commercial buildings right after the bombing is available in the Statistical Abstract of Hiroshima (*Hiroshima shisei youran*). For 1953, we exploit the Survey on the Daytime Population of Hiroshima (*Hiroshima-shi chukan jinko chosa*), where we assume that the daytime population approximates employment. From 1957 to 1975, we use the Business Establishment Statistical Survey (*Jigyousho toukei chousa*).¹⁶ Based on these data, we approximate block level employment every five years from 1950 to 1975. Panel (b) of Figure 2 shows employment in Hiroshima over time. Total employment dropped significantly in 1945 after the bombing, but increased again post-war, and the number of workers employed in the central area recovered to its pre-war level. The share of employment in the city center rose throughout the post-war period, implying an increased concentration of employment.

Commuting and Transportation Networks We use trip-level microdata from the 1987 Hiroshima City Person-Trip Survey to analyze commuting patterns. The data captures the workplace, residence, and representative travel mode for each commuting trip. We also collect and digitize road networks, bus networks, and train networks in Hiroshima and compute bilateral travel time between blocks for different modes: walk, bike, car, bus, and train. Although Hiroshima's public transportation networks were generally stable after the war, there were some changes, notably the discontinuation of the *Ujina* line in 1966.¹⁷ To address this, we use the public transportation networks of 1950 for years prior to 1966 and those of 1987 for later years.

Location Characteristics We collect various information on the location characteristics of each block. In particular, we exploit data on altitude, ruggedness, soil condition, geographical coordinates, distance to the pre-war CBD, distance to train stations, distance to Hiroshima port (*Ujina*

¹⁵While we do not distinguish employment by sector, available evidence suggests that the locations of manufacturing and service sector employment are highly correlated in Hiroshima (see Figure A.5 in the Appendix). We also abstract from agricultural employment. Even in 1950, when agricultural employment was large in the Japanese economy, the Population Census indicates that less than 10 percent of workers in Hiroshima City engaged in agriculture.

¹⁶For 1953–1963, employment is reported using less geographically granular units than blocks. We address this by combining the best available block-level information to approximate the block-level employment distribution. When employment data is unavailable but establishment data is available, we follow Ahlfeldt et al. (2015) and assume that the number of establishments is proportional to employment.

¹⁷The *Hijiyama/Minami* line and *Eba* line) opened in 1944 for military purposes, and these lines have been maintained after the war. These lines improved transportation access in the outskirts.





Note: The figures show total population and employment within the entire city and within 1 kilometer of the CBD (left axis), as well as the shares of population and employment within 1 kilometer of the CBD (right axis).

port), distance to bodies of water, and distance to cultural assets for each block. Distances are calculated using the centroid of each block shape.

3 Reduced-form Evidence

In this section we analyze population density to illustrate the pattern of destruction and recovery of the city structure of Hiroshima. Subsection 3.1 describes how the atomic bombing destroyed central Hiroshima and how the city subsequently recovered to its pre-war city structure. In Subsection 3.2, we formalize this recovery result in a regression analysis. We also find that the recovery tendency is robust to controlling for observable prominent location characteristics, which refer to characteristics of a location that directly affect amenities and productivity independently of the local density of economic activities (e.g., altitude, access to natural water).

3.1 Descriptive Evidence on the Destruction and Recovery of City Structure

In Figure 3 we non-parametrically plot population density within Hiroshima by distance to the CBD, where we normalize the total population of the city to 100,000 each year to facilitate comparisons of the inner-city structure over time. The figure shows that the city structure of Hiroshima was completely changed by the atomic bombing but quickly recovered to the pre-WWII city structure. In 1936, the city had a typical monocentric structure: the city center had the highest population density and density fell as one moved away from the center. This monocentric pattern was completely reversed by the atomic bomb hitting the densely populated city center. Figure 3 shows that, after the bombing, the city center was wholly destroyed and consequently had the lowest population density in the city. In contrast, areas two kilometers away from the center, which avoided total destruction (see Figure 1b), became the most crowded places in the city. Areas further away from the city center also experienced significant increases in population density as many survivors from the center escaped to the outskirts.



Figure 3: Population density by distance to city center

Note: The figure shows the local polynomial regression of log population density on distance to the CBD for different years. To eliminate the effect of changes in the total population, we normalize the total population each year to 100,000. The predicted population distribution of 1950 is computed based on the 1936 population distribution, assuming that each block experienced annual population growth rate equal to the pre-war (1933–1936) rate.

Despite the "reversal" of the monocentric city pattern after the bombing, the monocentric structure had already re-emerged in 1950, just five years after the bombing. The rate of population recovery in the city center was remarkable. While the recovery from total destruction is qualitatively clear, the recovery may not have been perfect, as the concentration of population around the CBD appears to be less dense in 1950 than in 1936. However, this does not necessarily imply that the recovery was incomplete because the city center already had a slow rate of population growth prior to the war. This can be illustrated by comparing the actual population distribution in 1950 and a predicted 1950 distribution, based on extrapolating pre-war population growth trends from 1936 to 1950. The next section formalizes this recovery result via a regression analysis, which allows us to consider the statistical significance of our findings and incorporate various location characteristics as controls.

3.2 Regression Analysis of the Recovery of Central Hiroshima

We now analyze the magnitude of the recovery at the spatially granular level of blocks. Note that recovery implies that the set of blocks that lost more population due to the atomic bombing grew faster in the post-war period. We operationalize this idea with the following regression model:

$$\ln\left(\frac{\text{Popdens}_{i,t}}{\text{Popdens}_{i,1945}}\right) = \gamma \ln\left(\frac{\text{Popdens}_{i,1945}}{\text{Popdens}_{i,1936}}\right) + \eta X_i + v_i,\tag{1}$$

where *i* is the block, *t* is the post-war year, e.g. 1950, X_i is the vector of location characteristics, and v_i is the error term. We regress the post-war log change in population density on the log change in population density from 1936 to 1945, reflecting the damage from atomic bombing.¹⁸ The estimated coefficient γ represents the degree of recovery back to the pre-war city structure.¹⁹ If $\gamma = 0$, the population density lost during the war did not recover in the post-war period. This would imply that the shock of atomic bombing has permanent effects on the city population distribution. In contrast, if $\gamma = -1$, the lost population density completely recovered in the post-war period, so the shock had only temporary effects on population distribution. To check robustness, we also consider an alternative regression specification at the end of this section.

In some specifications we control for location characteristics X_i such as altitude and distance to bodies of water, allowing us to investigate how the degree of recovery γ changes after conditioning on location characteristics. We interpret these regressions as capturing the correlation between population density lost during the war and post-war population growth, either unconditional or conditional on location characteristics X_i . It does not necessarily have a causal interpretation.

Panel (a) of Figure 4 illustrates the relationship between the wartime and post-war population density growth rates for each block as of 1950, along with the regression line. The fitted line is somewhat less steep but already close to a slope of -1, implying a strong resurgence among destroyed areas just five years after the bombing. Panel (b) of Figure 4 demonstrates that a similar result is obtained when examining the population distribution in 1960, suggesting that the recovery was essentially complete by 1950. Therefore, for brevity, we confine our regression analysis to the recovery from 1945 to 1950.

Column (1) of Table 1 provides the regression result for growth of population density in 1950, as depicted in Panel (a) of Figure 4. The coefficient is -0.712, which is statistically distinguishable

¹⁸We use 1936 population density because it is the closest observation we have before 1945. In our data, population density in 1945 is measured in November 1945 and is positive for all blocks.

¹⁹Since we include a constant in X_i , γ is invariant to population level in year t. Put differently, the coefficient γ captures the degree of convergence to the pre-war population distribution, where the total city population is normalized to the pre-war one.

from zero and the null hypothesis of complete persistence is rejected.²⁰ Although we can also statistically reject the null hypothesis of purely temporary shocks ($\gamma = -1$), the results suggest a strong recovery within just five years.





Note: The figures plot changes in the log of population density from 1945 to 1950 or 1960 with those from 1936 to 1945, which are largely driven by the atomic bombing. Each circle represents a block, where the size of the circle is proportional to the population density in 1936. We plot the (unweighted) linear fit between these two variables (solid line) as well as a line of slope -1 (dashed line), which would be obtained if population changes from the bombing were completely reversed in the post-war period.

Accounting for Location Characteristics We have documented the quick recovery of Hiroshima's city center. We now examine the extent to which the recovery of the city center can be explained by observed location characteristics that could directly affect amenities and productivity independently of the local density of economic activities, such as altitude and access to natural water.

Before proceeding with the regression analyses, we discuss heuristically why the locational advantage of central Hiroshima may not account for its resurgence. Following Krugman (1993), we consider two types of location characteristics: natural location characteristics (known as "first nature") and built location characteristics (known as "second nature"). For the first, natural conditions within the city, such as altitude and distance to water, are homogeneous because our geographic scope is limited and most of the city lies within 6 kilometers of the city center.²¹ For

²⁰The spatial autocorrelation of error terms is unlikely to affect this conclusion. First, using Conley's (1999) standard error to accommodate the spatial autocorrelation within 1 kilometer of each observation does not alter the statistical significance. Second, the residuals do not have a statistically significant correlation with distance from the CBD. Similarly, we also find that the potential spatial autocorrelation of errors would also be inconsequential after controlling for location characteristics.

²¹In particular, the majority of Hiroshima is located in the delta of the Ota river, characterized by flat terrain with

	(1)	(2)	(3)	(4)	(5)	(6)
	(Change in l	og populat	ion density	1945-1950	
Change in log population density 1936–1945 (γ)	-0.7124^{a}	-0.7443^{a}	-0.8004^{a}	-0.8383^{a}	-0.8307^{a}	-0.9034 ^a
	(0.0271)	(0.0343)	(0.0424)	(0.0560)	(0.0533)	(0.0465)
<i>p</i> -value from testing $\gamma = -1$	0.000	0.000	0.000	0.004	0.002	0.040
Natural location characteristics (first nature)		Yes		Yes	Yes	Yes
Built location characteristics (second nature)			Yes	Yes	Yes	Yes
Pre-war trends in population					Yes	
Within 3 km of the city center						Yes
Number of blocks	174	174	174	174	174	158
R-squared	0.809	0.828	0.823	0.846	0.848	0.859

Table 1: Change in population density and war-time damage

Note: We report the OLS estimates of equation (1). Natural location characteristics consist of log distance to nearest water, altitude and its square, ground slope and its square, geographical coordinates (latitude, longitude, and their interaction), and a dummy for bad soil conditions. Built location characteristics consist of log distance to the nearest station, log distance to Hiroshima port (*Ujina* port), log distance to the nearest cultural asset, and the initial housing stock condition (the fraction of moderately-destroyed or intact buildings). In column 5, we also control for the pre-war (1933-1936) population growth rate and its square. In column 6, we confine the sample to blocks within 3 kilometers of the city center. We report the *p*-value from testing the null $\gamma = -1$, meaning that the population density converged back to the 1936 city structure. Heteroskedasticity-robust standard errors in parentheses. ^{*a*} indicates significance at the 1 percent level.

the second, the built advantages of central Hiroshima were substantially damaged by the bombing. The city center of Hiroshima, areas around *Hacchobori* and *Kamiya-cho*, is located next to Hiroshima castle, a historical amenity which had been a symbol of the city since the samurai period. The center was also adjacent to the former center, called *Nakajima-cho*, which developed during the samurai period due to its convenient access to the castle and water transportation. These advantages were lost following the bombing. Hiroshima castle was completely destroyed as was *Nakajima-cho*. Although the city center may have retained some transportation advantage, jobs and other economic amenities would have been eliminated as central neighborhoods were completely destroyed and other areas of the city likely enjoyed better conditions after the bombing, as can be seen in other Japanese cities.²²

We now use our regression model (1) to formally assess the role of locational advantages. Specifically, we control for the observable characteristics of each block. If the recovery were primarily driven by the attractive location characteristics of the destroyed areas, then γ would approach zero, such that the recovery tendency is no longer observed after conditioning on these

loose soil. The flat terrain was an important reason why the US chose Hiroshima as the target of the bombing (see footnote 6). Moreover, much of the city close to water, as the city is cut through by many branches of the *Ota* and faces the sea to the south.

²²The areas around Hiroshima station also provided convenient access to transportation but experienced much less destruction from the bombing, which could have made Hiroshima station the potential new center of Hiroshima. In Japan, some large cities did see their centers shift after the war (e.g., Yokohama, Kobe). In Yokohama, Takano (2023) documents that the city center moved to an area with transportation advantages after the requisition of the former city center by the U.S. Army for nearly ten years.

location characteristics. For natural location characteristics, we control for distance to nearest water, altitude and ruggedness, an indicator of bad soil conditions, and geographic coordinates.²³ As built location characteristics that can be considered as given right after the bombing, we control for distance to the nearest train station as of 1950, distance to Hiroshima port, distance to the nearest cultural asset, and the quality of housing stock after the bombing.²⁴

Columns (2)–(4) of Table 1 present the regression results when controlling for location characteristics. Column (2) controls for natural conditions. We find that natural characteristics do not account for the recovery as the coefficient γ actually moves closer to negative one. Column (3) controls for built conditions, again finding that γ gets closer to negative one relative to Column (1). In Column (4) we control for both natural and built location characteristics. We estimate γ = -0.83, which is even closer to negative one. This conclusion remains in Column (5) when controlling for pre-war trends in population. Note that in Columns (1)–(5), we can also reject the null of perfect path independence ($\gamma = -1$), meaning that the post-war city structure is somewhat different from the pre-war one. Nevertheless, we consistently find γ closer to negative one than zero, meaning that the city structure exhibited a strong recovery tendency. Moreover, observed location characteristics do not explain the recovery, as the degree of recovery γ becomes stronger after conditioning on observed location characteristics.

Column (6) of Table 1 replicates Column (4) while restricting our sample to blocks within 3 kilometers of the city center. Within such a small area, it is harder to attribute the recovery of the destroyed areas to unobserved fundamentals because location characteristics are likely more homogeneous.²⁵ Even within such a small homogeneous sample, we find that γ gets even closer to negative one relative to Column (4). This result suggests that unobserved location advantages of the destroyed areas may not be the main driver of the recovery. Taken together, our results suggest that the recovery was not driven by advantageous location characteristics of the destroyed areas.

Testing Recovery via Alternative Specification Besides our main specification (1), we use an alternative regression specification where we regress the logarithm of population density in 1950 on the logarithm of population density in 1936 and 1945 and observed location characteristics:

$$\ln \text{Popdens}_{i,t} = \eta X_i + \gamma_1 \ln \text{Popdens}_{i,1945} + \gamma_2 \ln \text{Popdens}_{i,1936} + v_i \tag{2}$$

²³Note that considering potential radioactive contamination would, if anything, reinforce the main finding of our reduced-form analysis that the city center recovered. Since radioactive contamination is a "bad" that makes the city center less attractive, failing to control for it would underestimate the strength of the recovery.

²⁴We measure the quality of housing stock by the fraction of moderately destroyed or intact buildings. As an additional control for housing conditions, we use the number of public housing units constructed between 1945 and 1950 in a robustness check. This has little impact on the regression results (see Appendix B.2 for details).

²⁵Table A.2 in the Appendix suggests that the standard deviations of observed location characteristics are smaller for blocks within 3 kilometers of the city center. A similar idea has been invoked in Schumann (2014) in a different context.

This alternative specification allows us to address two concerns.²⁶ First, measurement error in the 1945 population may introduce a negative bias into the coefficient γ in equation (1). Specifically, a positive measurement error reduces the outcome variable while increasing the explanatory variable in equation (1), but not in equation (2). Second, by regressing the 1950 population density on location fundamentals (X_i) exclusively, we can determine the extent to which city structure can be explained by location fundamentals. If fundamentals account for most of the variation in population density within a city, then the recovery tendency we found in Table 1 may not be economically important, and vice versa.

Table B.1 in the Appendix presents the results of estimating (2). Columns (1)–(3) of Table B.1 address the first concern of measurement error. We find that the 1936 population density, i.e. the pre-war city structure, is much more strongly associated with the 1950 population density than the 1945 population density, immediately after the bombing. Notably, the 1945 density is no longer significant when built location characteristics are controlled for. As in Column (6) of Table 1, Column (4) of Table B.1 restricts the sample to areas within 3 kilometers of the CBD to limit heterogeneity in location fundamentals. Reassuringly, the above conclusion remains valid in Column (4). Overall, the findings in Table 1 are robust to this alternative specification, and the city structure exhibits a strong recovery tendency.

Next, in Columns (5) and (6), we regress the 1950 population density solely on location fundamentals to determine the extent to which they account for the city structure. While there is some evidence suggesting a correlation between distance to nearest station and water and lower population density, the ability of the specification containing only fundamentals to explain population density is substantially worse than that of our baseline specifications in Columns (3) and (4), which include 1936 and 1945 population density.²⁷ Therefore, the city structure tends to recover to the pre-war city structure, above and beyond the tendency that blocks with advantageous location fundamentals achieve higher population density. This result aligns with our conclusion drawn from Table 1, which suggests that the recovery was not primarily due to advantageous location fundamentals and that other forces likely induced the recovery.

3.3 Robustness

We provide further reduced-form evidence on the recovery of the city center by exploiting other information and specifications. In addition, we show that recovery to the pre-war city structure was also observed in Nagasaki, the second city hit by atomic bombing, and discuss the implications. The results are in Appendix B.

²⁶This specification is equivalent to (1) if we impose that the sum of two coefficients ($\gamma_1 + \gamma_2$) equals one.

 $^{^{27}}$ Indeed, the R-squared of Columns (3) and (4) is substantially larger than that of Columns (5) and (6)

Recovery of Employment Distribution In Section B.2 of the Appendix we analyze the impact of the atomic bombing on the distribution of employment. We find that the regression results are very similar to those in Table 1 for the population distribution. Recovery occurred not only for the residential population but also in the spatial distribution of commercial activity.

Recovery of Land Price Distribution While we do not have comprehensive land price data during our sample period, we are able to measure the location with the highest land price in the city, which could be interpreted as the center of the city. In both 1931 and 1959, the highest land price was observed near *Hacchobori*, the city center both before and after the war.²⁸ Thus, the resurgence of central Hiroshima is also observed in land prices.

Characteristics of Neighboring Blocks In Section B.2 of the Appendix, we consider the possibility that the post-war population growth rate of a block may depend not only on its own characteristics, but also on the characteristics of neighboring blocks. To consider the characteristics of neighbors, we adopt the so-called "SLX model" from the spatial econometrics literature (Halleck Vega and Elhorst 2015) and add spatial lags of the following three neighborhood characteristics to our main regression (1): (i) the log change in population induced by the bombing; (ii) the location characteristics; and (iii) the population distribution right after the war, which is meant to capture market access after the bombing.²⁹ Table B.3 shows that including these spatial lag variables has only a modest impact on our regression results.

The Recovery of Nagasaki In Section B.3 of the Appendix we examine population data for Nagasaki, the second city destroyed by an atomic bomb. As in Hiroshima, the damage in Nagasaki was catastrophic: around 70,000 people were killed and almost all buildings within 2 kilometers of the epicenter were wholly destroyed (Nagasaki City Government 1977). However, in Nagasaki, the atomic bomb hit the outskirts of the city (see Figure B.1a). This is in contrast to Hiroshima, where the atomic bomb hit the city center.³⁰ Despite this difference, recovery to the pre-war city structure was also observed in Nagasaki. Figure B.1b shows a fitted line from estimating equation (1) for Nagasaki. Our coefficient of interest is around –0.88 and statistically indistinguishable from –1, suggesting the complete recovery of the population distribution in Nagasaki. This similar pattern of recovery in Nagasaki offers two additional implications. First, it bolsters the external

²⁸The block containing the plot with the highest land price is the *Horikawa-cho* block adjacent to the *Hacchobori* block. Other available evidence on pre-war land prices suggests a similar conclusion (Nozawa 1934; Hayakawa and Nakaouji 1965).

²⁹In particular, the market access term addresses the possibility that the city center could recover as the "donut hole of the city": it might still have relatively good market access thanks to its central location despite its destruction.

³⁰Nagasaki was selected by historical coincidence. The US initially intended to bomb Kokura, but changed to the city center of Nagasaki due to weather conditions. The weather conditions also prevented an attack on the city center of Nagasaki, and consequently the bomb was dropped on the outskirts of Nagasaki. See https://www.peace-nagasaki. go.jp/abombrecords/b020101.html (last accessed on October 28, 2023).

validity of our Hiroshima recovery result. Second, it may limit the importance of potentially lower development costs on large empty plots or creative destruction in our context because the center of economic activities could have shifted toward the completely destroyed periphery of Nagasaki if these factors had been crucial, but we do not find such a shift.³¹

Summary of Reduced-form Analysis Taken together, our reduced-form analysis has revealed that (i) the population distribution in the city recovered back to its pre-war state within five years after the bombing, and (ii) the recovery was not driven by the prominent location characteristics we have controlled for. There are two possible explanations for these findings. First, there could be some unobserved locational advantages in the destroyed city center that persisted through the bombing (e.g., scenic views). Second, people may have expected the recovery of the destroyed city center when making location choices, and the incentive to again live and work in the city center came from agglomeration forces due to expected high density as in the pre-war period. To analyze these possibilities, we develop and calibrate a quantitative spatial model that incorporates both agglomeration forces and unobserved location characteristics as potential explanations for the recovery.

4 Theoretical Framework

In this section we present a dynamic quantitative spatial model to understand the mechanisms of the recovery. To account for the impact of the atomic bombing on the dynamics of internal city structure during the recovery, the model incorporates forward-looking location choice, migration frictions, commuting, agglomeration forces, and heterogeneous location fundamentals.

First, individuals make decisions about their residence and workplace subject to migration frictions. They do so taking into account continuation values, defined by the expected future value of living and working in their chosen locations. Second, their residence and workplace are potentially different, which defines the equilibrium commuting patterns within the city. Finally, locations differ in productivity and amenities that are determined by both agglomeration forces and location fundamentals. Our model is the first tractable dynamic quantitative model of internal city structure that possesses these elements in a unified framework. Appendix C provides the details of the derivations.

Time is discrete and indexed by *t*. We consider a single city (Hiroshima City) embedded in a large economy (Hiroshima Prefecture or Japan). The city consists of a discrete set of locations rep-

³¹The limited importance of large empty lots or creative destruction in our context may be due to construction technology. While the demolition of old buildings facilitates high-rise buildings by providing a large vacant lot and may enhance productivity (Hornbeck and Keniston 2017), Japanese technology for high-rise buildings right after WWII was relatively limited and the shortage of construction materials made high-rise buildings even more unavailable (Hiroshima City Government 1971).

resented by C. We typically use subscripts n and n' to refer to the place of residence of a worker and subscript i and i' to refer to their place of work. The number of locations in the city N = |C|is fixed over time. The city is located within a larger economy that is modelled as a single location, denoted by o. Locations in the city correspond to blocks and are differentiated by fundamental productivity, amenities, land endowments, and geography. Fundamental productivity and amenities capture exogenous locational advantages for production and residence, respectively, and can change over time. The land endowment of each block is constant over time.

Individuals in the economy live for a finite time, T. The mass of the population in the larger economy is denoted by M and is exogenous. While we assume that M is time-invariant, the total population of the city changes over time through migration flows. This allows us to focus on the distribution of individuals within the city. Individuals are endowed with one unit of labor that is supplied inelastically and they are geographically mobile across locations in a city. They commute from their residential block to their workplace block subject to commuting costs. Individuals outside the city (o) are prohibited from commuting into the city for work. Production occurs in every location in the economy and firms produce a homogeneous final good that can be freely traded across locations. In each period, an individual may receive an opportunity to change their residence and workplace. Individuals receive such an opportunity with some exogenous probability. Given an opportunity to change locations, individuals choose their locations in a forward-looking way, taking into account the expected future values of living and working in particular locations. These forward-looking location choices allow us to characterize the transitions of the population and employment distributions in the city, which depend on agents' expectations about the future.

4.1 Production

Firms in the economy are competitive and produce a homogeneous final good. The production technology of a representative firm in location $i \in C$ is:

$$Y_{it} = A_{it}L_{it},\tag{3}$$

where Y_{it} is production in location *i*, A_{it} is productivity and L_{it} is employment in location *i* at time *t*. Productivity (A_{it}) is determined by fundamental productivity and employment density in the location:

$$A_{it} = a_{it} \left(\frac{L_{it}}{S_i}\right)^{\alpha},\tag{4}$$

where a_{it} represents the exogenous component of productivity and S_i is the area size of location i that is time-invariant. The parameter α controls the contemporaneous productivity agglomeration forces with respect to employment density, and a positive value of α implies higher employment

density increases productivity.³²

The homogeneous good is freely traded and chosen as the numeraire.³³ The zero-profit condition implies that the wage rate at location *i* in period *t* is $w_{it} = A_{it}$. Therefore, the wage rate in any particular location is a function of exogenous productivity and employment density in that location.³⁴

4.2 Preferences

Individuals live for finite periods, consume only a homogeneous tradable good, and inelastically supply one unit of labor. Individuals are hand-to-mouth, always spending their wage w_{it} in period t. Their period utility from living in location n and working in location i at period t is:

$$\ln u_{int} = \ln B_{nt} + \ln w_{it} - \ln \kappa_{int},\tag{5}$$

where B_{nt} is the common utility benefit from residential amenities at residential place *n* in period *t*, w_{it} is labor earnings in workplace *i*, and κ_{int} is the utility cost due to commuting from *n* to *i*.

The value of amenities in a residential place (B_{nt}) depends on the fundamental value of amenities and population density:

$$B_{nt} = b_{nt} \left(\frac{R_{nt}}{S_n}\right)^{\beta},\tag{6}$$

where b_{nt} is an exogenous component in the value of amenities for each location and R_{nt} is the population of location n in period t. In this specification, the elasticity of amenities with respect to population density (β) captures the strength of the net agglomeration effect in a residential place. Specifically, we capture net residential agglomeration forces through housing prices and consumption amenities. In Appendix C.5, we provide a microfoundation for this specification.³⁵

Outside of the city (o), individuals receive common utility u_{ot} in period t, which is exogenous in every period. Because u_{ot} governs the attractiveness of living in Hiroshima City relative to the outside economy, it captures aggregate shocks that affect the whole city of Hiroshima.

³²This is in line with numerous empirical findings (Ciccone and Hall 1996; Arzaghi and Henderson 2008) and microfounded by different mechanisms in agglomeration economies (Duranton and Puga 2004). In addition, this may also reflect congestion forces from land prices that are increasing in density. We do not explicitly include land in production for simplicity, as we do not observe block-level land prices in the data.

³³We can incorporate block-specific costs of transporting the homogeneous good to the outside world. Suppose that the numeraire produced in block *i* is exported to the outside world subject to iceberg trade cost: $\tau_i \ge 1$ units of goods must be shipped to sell one unit. This is isomorphic to the case with productivity A_{it}/τ_i in our model.

³⁴Given the linear technology and perfect competition, producers always earn zero profit. Thus, considering dynamic incentives does not change our arguments as long as firms correctly expect that future profits will always be zero. In addition, no firm has an incentive to enter or exit in any location.

³⁵We do not explicitly model consumption of land for simplicity since we do not observe block-level land prices in the data.

4.3 Forward-looking Location Choices

Workers are forward-looking in making migration decisions subject to the exogenous migration frictions. At the end of period t, share $\theta_t \in (0, 1)$ of workers in the economy can change their locations and share $1 - \theta_t$ of workers will remain in their current locations next period t + 1. If $\theta_t = 1$, all workers are able to change their location pairs. A low value of θ_t leads to stickiness in workers' location decisions.³⁶ Intuitively, due to the high fixed cost of moving, migration happens when an exogenous event arrives, such as job loss or life-cycle shocks. θ_t is interpreted as the probability of experiencing such an event.

When a worker obtains the opportunity to change their locations at the end of period t, they draw idiosyncratic shocks related to location choice in period t + 1. For an individual worker, the idiosyncratic shock is independently drawn from a time-invariant independent Type–I extreme distribution $F(\varepsilon) = \exp(-\exp(-(\varepsilon + \Gamma)))$ where Γ is the Euler-Mascheroni constant. At the end of period t, workers decide their residence and workplace for the next period considering the option value $\{V_{int+1}\}$ associated with each workplace and residence pair.

Consider a worker ω living in *n* and working in *i* at period *t*. When the worker can move to a different location pair next period, they solve the following location choice problem:

$$v_{int}(\omega) = \ln u_{int} + \max\left\{\rho V_{i'n't+1} + \sigma \varepsilon_{i'n't+1} ; \rho V_{ot+1} + \sigma \varepsilon_{ot+1}\right\}$$
(7)

for $t = 1, 2, \dots, T - 1$. $V_{i'n't+1}$ refers to the value function implied by choosing a different residence n' and workplace i' in period t + 1 and V_{ot+1} is the option value of choosing to live outside the city. $\rho \in (0, 1)$ is the discount factor governing the importance of the future values and σ is a positive constant governing the variance of the idiosyncratic shocks. An individual makes a forward-looking migration choice of residence and workplace at t + 1 given a path of the exogenous and endogenous variables. In particular, an individual correctly anticipates the path of the population distribution (R_{nt}) and employment distribution (L_{it}) that are endogenously determined in equilibrium. As we focus on migration within a city, we assume away bilateral mobility costs as they are likely sufficiently small and homogeneous relative to inter-city migration costs.³⁷

With the idiosyncratic shocks following a Type–I extreme value distribution and migration frictions, we can express the option value of living in n and working in i in period t by:

$$V_{int} = \ln u_{int} + (1 - \theta_{t+1})\rho V_{int+1} + \theta_{t+1}\sigma \ln \left[\sum_{i' \in \mathcal{C}} \sum_{n' \in \mathcal{C}} \exp(\rho V_{i'n't+1})^{1/\sigma} + \exp(\rho V_{ot+1})^{1/\sigma} \right],$$
(8)

³⁶This Calvo-style migration friction is also adopted in other recent quantitative spatial models to capture the persistence of migration decisions (Caliendo, Dvorkin, and Parro 2019 Section 5.3; Heblich, Trew, and Zylberberg 2021). This approach is attractive in a setting such as ours in which bilateral migration flows are unobserved.

³⁷Unlike Caliendo, Dvorkin, and Parro (2019) we do not observe bilateral migration flows. This is likely inconsequential as Gechter and Tsivanidis (2023) estimates that in a within-city setting, the fixed cost of moving is substantially larger than the moving cost that increases with moving distance.

for t = 1, 2, ..., T - 1. The first term is the current utility from residence *n* and workplace *i*. The second term is the expected value of staying at the same location pair next period when no migration opportunity realizes, with probability $1 - \theta_{t+1}$. The third term is the expected value when a worker is able to change their location pair, with probability θ_{t+1} .

For workers residing outside of the city, their option value for t = 1, 2, ..., T - 1 is:

$$V_{ot} = \ln u_{ot} + (1 - \theta_{t+1})\rho V_{ot+1} + \theta_{t+1}\sigma \ln \left[\sum_{i' \in \mathcal{C}} \sum_{n' \in \mathcal{C}} \exp(\rho V_{i'n't+1})^{1/\sigma} + \exp(\rho V_{ot+1})^{1/\sigma}\right].$$
(9)

When workers have an opportunity to migrate, they can choose any location pair. Therefore the last term in the value function is the same as in equation (8). For the last period t = T, equations (8) and (9) are written as $V_{inT} = \ln u_{inT}$ and $V_{oT} = \ln u_{oT}$ because future considerations are absent.

Using our assumption that the idiosyncratic shocks are independent and follow a type-I extreme value distribution $F(\varepsilon)$, we derive the share of workers that live in *n* and work in *i* in the next period t + 1 when they have a migration opportunity:

$$\lambda_{int+1} = \frac{\exp(V_{int+1})^{\rho/\sigma}}{\sum_{i'\in\mathcal{C}}\sum_{n'\in\mathcal{C}}\exp(V_{i'n't+1})^{\rho/\sigma} + \exp(V_{ot+1})^{\rho/\sigma}}, \quad i,n\in\mathcal{C}.$$
 (10)

This probability λ_{int+1} characterizes the location dynamics of workers in the city for period t + 1. Workers choose their pair of residence and workplace, correctly anticipating future changes in commuting costs, wages, and residential amenities. Since there is no residence-workplace specific migration cost, equation (10) applies to all workers with a migration opportunity in period t. In addition, the share of workers that live outside of the city in period t + 1 conditional on being able to change their location pair is given by probability $\lambda_{ot+1} = 1 - \sum_{i \in C} \sum_{n \in C} \lambda_{int+1}$.

Using these choice probabilities for workers, we can express the mass of workers in the city who live in n and work in i in period t + 1 as:

$$L_{int+1} = (1 - \theta_{t+1})L_{int} + \theta_{t+1}\lambda_{int+1}M.$$
(11)

This is the number of commuters within the city. On the right-hand side, the first term is equal to the number of commuters who retain the same workplace and residence from the last period, and the second term is the total number of workers who either move in from outside of the city or change location pairs within the city. Since the commuting market clears, the mass of workers in workplace *i* becomes:

$$L_{it+1} = (1 - \theta_{t+1})L_{it} + \theta_{t+1} \left[\sum_{n \in \mathcal{C}} \lambda_{int+1}\right] M,$$
(12)

where the mass of workers in workplace i is the sum of workers who have no opportunity to change locations and those who join workplace i in period t. Analogously, the mass of workers residing in n becomes:

$$R_{nt+1} = (1 - \theta_{t+1})R_{nt} + \theta_{t+1} \left[\sum_{i \in \mathcal{C}} \lambda_{int+1} \right] M.$$
(13)

Lastly, the total population in the city in period t + 1 is given by $L_{t+1} = \sum_{i \in C} \sum_{n \in C} L_{int+1}$.

Conditional on the wage variation and exogenous location characteristics, the mobility of workers in our model is controlled by the parameter of Calvo-style stickiness θ_t and taste shocks σ . We emphasize that they have different interpretations. Calvo-style migration frictions capture the immobility of workers even if they would like to change their locations. Intuitively, this reflects any constraint that prevents workers from relocating. In contrast, the dispersion of taste shocks captures the individual valuation attached to the location pair and controls the degree of sorting in response to utility differences. In the present model, we introduce both migration frictions and idiosyncratic shocks to capture both mobility constraints and the sorting of workers into their residence and workplace choices. We discuss how we can identify these two parameters from the data in Subsection 5.1 below.

4.4 General Equilibrium

We now define a forward-looking competitive equilibrium in this economy. The economy starts with the initial distributions of population (R_{n0}) and employment (L_{i0}). The exogenous variables of the model are block-level fundamental productivity (a_{it}) and amenities (b_{nt}), the sizes of blocks (S_n), bilateral commuting costs (κ_{int}), the degree of worker location stickiness (θ_t) and utility outside the city (u_{ot}). The economy-wide parameters in the model are the agglomeration forces in productivity (α), agglomeration forces in amenities (β), the discount factor of workers (ρ), the variance of idiosyncratic shocks in location choices (σ) and the mass of workers in the economy (M). Then, a forward-looking equilibrium is defined as follows:

Definition 1. Given the exogenous variables of the model and economy-wide parameters, a forwardlooking equilibrium is characterized by the sequences of wages $\{w_{it}\}$, population $\{R_{nt}\}$, employment $\{L_{it}\}$, and value functions associated with location choices $\{V_{int}\}$ such that (i) the value functions of workers for their location choices $\{V_{int}, V_{ot}\}$ satisfy (8) and (9) with $V_{inT} = \ln u_{inT}$ and $V_{oT} =$ $\ln u_{oT}$ for the last period T, (ii) the commuting market clears in the city and the masses of workers in workplace and residential locations are given by (12) and (13), and (iii) firms maximize their profits and the zero-profit condition leads to a wage rate equal to (4).

Since productivity and amenities evolve with employment density and population density, we can summarize the forward-looking equilibrium by population, employment, and the value function adjusted by the value of living outside of the city. Equations (4), (8), (9), (12), and (13) constitute $N^2 + 3N$ equations for each t, which can be solved for $N^2 + 3N$ endogenous variables for each t. Location choices are based on current real income but also the option values associated with each pair of locations, and they determine the future path of location choices taking into account future shocks.

We define a steady-state equilibrium for the economy as one where the population and employment distributions are constant. The steady state equilibrium in this economy exists and is unique when the net agglomeration forces are small for both productivity and amenities. We summarize these results in the following proposition:

Proposition 1. (i) Given the initial state and exogenous factors, a forward-looking competitive equilibrium such that, for all periods $t = 1, 2, \dots, T$, $R_{nt} \ge (1 - \theta_t)R_{nt-1}$ and $L_{it} \ge (1 - \theta_t)L_{it-1}$, exists; (ii) A steady-state equilibrium exists when $\alpha \ne \sigma/\rho$ and $\beta \ne \sigma/\rho$; (iii) Sufficient conditions for the existence of a unique steady state are negative net agglomeration forces: $\alpha \le 0$ and $\beta \le 0$.

Appendix C.3 provides the proof. While Proposition 1 (i) and (ii) show the existence of an equilibrium and a steady state, there may be multiple steady states and multiple equilibrium paths. Proposition 1 (iii) shows that the steady state is unique if agglomeration forces (α , β) are negative, which implies that the net dispersion forces dominate the agglomeration forces both in productivity and amenities. In this case, the economy will converge to this unique steady state in the long run, although there could be multiple paths toward the steady state. In other cases, there can be multiple steady states, which can happen when net agglomeration forces are positive according to Proposition 1 (iii). After a shock that affects the initial condition, both the initial conditions and expectations about the future distribution of population and employment matter in determining which steady state or path realizes (Krugman 1991; Matsuyama 1991).

In our calibration, we solve the model backward for the observed changes in population and employment. We do not require that the economy is exactly in the steady state in the last period T, but we assume that it is sufficiently close to the steady state so that the commuting gravity equation approximately holds, which we estimate in our calibration. Our calibration does not require the uniqueness of the steady state nor a unique path to the steady state because it relies only on the observed equilibrium. This feature allows us to calibrate the model when there are multiple steady states so that different expectations may lead to different steady states. When we undertake counterfactuals, we explicitly acknowledge the potential for multiple equilibria.

5 Quantitative Analysis

Our goal in this section is to show how the present model can be matched to the observations in Hiroshima. Our quantification proceeds in three steps which we discuss in turn. Appendix D presents further details on the calibration.

In Subsection 5.1 we first obtain commuting costs (κ_{int}) by estimating a model of travel mode choice. Our model accommodates two aspects of migration frictions. We calibrate each using different information. The dispersion of idiosyncratic taste shocks (σ) is calibrated based on the calibrated discount factor (ρ) and the commuting elasticity (ρ/σ) estimated by a gravity equation for commuting. We infer the stickiness (θ_t) from additional data on the share of people who remain in the same residence over time. The outside utility (u_{ot}) is chosen to match the observed total population of the city.

Given the parameters, in Subsection 5.2, we leverage the structure of the model to back out the composites of amenities and productivity that rationalize the observed population and employment changes over time. Intuitively, changes in population and employment by block allow us to invert the option values associated with each location. These option values reflect the attractive-ness of each location as a residence or workplace, which is a composite of location fundamentals and agglomeration forces.

In Subsection 5.3, we estimate the key parameters that govern the strength of the agglomeration forces in productivity (α) and amenities (β). We first recover the unobserved fundamentals in productivity and amenities based on the recovered option values and variations of population density and employment density over time. For these fundamentals, we then define the moment conditions and estimate the agglomeration force parameters. In the estimation, we use the location choice data from 1955 to 1975. In Subsection 5.4 we discuss the robustness of our estimated values for agglomeration forces.

Having fully quantified our model, in Subsection 5.5 we investigate how well our model fits the observed changes in population and employment distributions in the recovery period. To this end, we first use the location choice data for 1950, which is not used for calibration, to back out the locational advantages in the recovery period. We then decompose these advantages into two components: (i) advantages in productivity and amenities explained by the model, and (ii) structural residuals in productivity and amenities. We demonstrate that our model predicts the central recovery only with the first model-based component.

5.1 Step #1: Parameter Calibration (ρ , σ , κ_{int} , θ_t , u_{ot})

Travel mode choice and commuting costs (κ_{int}) To estimate commuting costs, we extend the model to incorporate choice of travel modes following Ahlfeldt et al. (2015) and Tsivanidis (2022). There are five modes of transportation: walk, bicycle, car, bus, and train. In each period, a worker chooses the mode of transportation that minimizes the realization of observed and idiosyncratic travel costs, given their workplace and residence. We assume that the idiosyncratic travel cost follows a Gumbel distribution with two nests: (i) public modes: walk, bus and train; and (ii)

private-vehicle modes: bicycle and car. We estimate this nested discrete choice model of travel mode by exploiting the 1987 Hiroshima City Person Trip Survey and compute the expected commuting cost for two types of workers who may or may not use cars.³⁸ We then estimate the overall expected travel cost for residence n and workplace i before the realization of the idiosyncratic travel costs, using information on the car ownership rate in Japan in different years. We discuss the details in Appendix D.1.

Commuting gravity (ρ/σ) We suppose that the economy approximately reaches a steady state in the last period and estimate the commuting elasticity of workers using the 1987 Hiroshima City Person Trip Survey.³⁹ Plugging the average commuting time in 1987 from above into the equilibrium commuting pattern in the steady state yields the gravity equation:

$$\ln L_{in} = -\frac{\rho}{\sigma}\bar{c}_{in} + \phi_i + \eta_n + \chi, \qquad (14)$$

where \bar{c}_{in} is the log bilateral commuting costs determined by travel time, ϕ_i and η_n are workplace and residence indicators and χ is a constant. ρ/σ corresponds to the commuting elasticity with respect to commuting cost in our model, which is decreasing in σ (the dispersion parameter of the idiosyncratic shock) and increasing in ρ (the discount factor). Lower σ and higher ρ imply a higher sensitivity of migration decisions to utility differentials. We estimate (14) using Pseudo-Poisson Maximum Likelihood to allow for heteroskedasticity and zero bilateral commuting flows for some pairs. Our baseline parameter estimate of ρ/σ is 8.019, which is close to estimates of the elasticity of commuting flows with respect to commuting costs in Dingel and Tintelnot (2020). In the following, we set ρ/σ to be 8 for all t. See Appendix D.1 for the detailed estimation results.

Discount factor (ρ) We assume that the annual discount rate is 8.5 percent. This value is consistent with the discount rate widely used in the context of developing countries (e.g., Garcia-Cicco, Pancrazi, and Uribe 2010), which is consistent with the relatively low GDP per capita of Japan right after the war.⁴⁰ Since one period in our calibration corresponds to five years, we set $\rho = (1/1.085)^5 \simeq 0.66$.

Migration frictions (θ_t) Individuals can change their residence and workplace in period *t* with probability θ_t . We assume that people change their residence when obtaining a migration opportunity and match this migration friction parameter to the probability that people change their residence during five years, the length of one period in calibration. The 1960 Population Census reports that around 86 percent of people stayed in the same residence from the prior year. Thus, we set the parameter $\theta_t = 1 - (0.86)^5 \simeq 0.53$ for all $t \ge 1955$.⁴¹

³⁸When a car is unavailable, the nest of private vehicle modes is reduced to a single choice (bicycle).

³⁹Alternatively, we can suppose that individuals can always migrate ($\theta_t = 1$) after the last period.

⁴⁰Japan's GDP per capita in 1950 was less than one-fifth that of the U.S.

⁴¹Although in a different context, this value is very close to the value used in Heblich, Trew, and Zylberberg (2021).

Utility outside the city (u_{ot}) We set the outside utility (u_{ot}) for each period to match the total population of Hiroshima City. Formally, we choose the outside value to match the observed total population in the city, \mathbb{M}_t . The model predicted population of the city is $(1 - \lambda_{ot})M = \mathbb{M}_t$, where λ_{ot} is the probability of choosing to live and work outside the city computed in the model. Since location choice is independent of the outside utility conditional on living in the city, the value of the outside option only affects the total population of Hiroshima in our model so we can focus on analyzing the internal city structure.

5.2 Step #2: Inversion of the Option Values

When individuals are forward-looking, their location choices depend on current real income and the option value associated with each location. In this step, we back out the option values by leveraging the population and employment dynamics of the model. Specifically, for $t = 1, 2, \dots, T - 1$, the option value of location n as a residential place can be summarized by the continuation value of amenities in the location:

$$\Xi_{nt} = b_{nt} \left(\frac{R_{nt}}{S_n}\right)^{\beta} \prod_{\tau=t+1}^{T} \left[b_{n\tau} \left(\frac{R_{n\tau}}{S_n}\right)^{\beta} \right]^{\prod_{s=t+1}^{t} \rho_s (1-\theta_s)}.$$
(15)

Analogously, the option value of location *i* as a workplace can be written as:

$$\Omega_{it} = a_{it} \left(\frac{L_{it}}{S_i}\right)^{\alpha} \prod_{\tau=t+1}^{T} \left[a_{i\tau} \left(\frac{L_{i\tau}}{S_i}\right)^{\alpha}\right]^{\prod_{s=t+1}^{\tau} \rho_s(1-\theta_s)}.$$
(16)

These option values express the attractiveness of each location as a residence and workplace. They are a composite of amenities and productivity that include both fundamental amenities (b_{nt}) and productivity (a_{it}), and the agglomeration forces from future population and employment density.

When $\theta_t = 1$, all workers can change locations every period and therefore the future values of their choices are independent of current location choices. In contrast, rare migration opportunities (small θ_t) lead to more weight placed on the future evolution of amenities and productivity since workers are less likely to change their locations. In sum, these option values reflect the value of amenities and productivity for each location when workers choose locations in a forward-looking way.

Equations (12) and (13) imply that the option values (Ξ_{nt} , Ω_{it}) satisfy the following equations:

$$R_{nt} - (1 - \theta_t)R_{nt-1} = \sum_{i \in \mathcal{C}} \frac{K_{int} \Xi_{nt}^{\rho/\sigma}}{\sum_{n' \in \mathcal{C}} K_{in't} \Xi_{n't}^{\rho/\sigma}} (L_{it} - (1 - \theta_t)L_{it-1}),$$

$$L_{it} - (1 - \theta_t)L_{it-1} = \sum_{n \in \mathcal{C}} \frac{K_{int} \Omega_{it}^{\rho/\sigma}}{\sum_{i' \in \mathcal{C}} K_{i'nt} \Omega_{i't}^{\rho/\sigma}} (R_{nt} - (1 - \theta_t)R_{nt-1}),$$
(17)

where K_{int} summarizes current and future commuting costs (see equation D.3 in the Appendix for the definition). Intuitively, equation (17) states that the number of residents that actively choose to live in block *n* for period $t (R_{nt} - (1 - \theta_t)R_{nt-1})$ is written as the sum of the products of the number of workers that actively choose to work in block *i* for period $t (L_{it} - (1 - \theta_t)L_{it-1})$ and their conditional residential choice probabilities for location $n (K_{int} \Xi_{nt}^{\rho/\sigma} / (\sum_{n' \in C} K_{in't} \Xi_{n't}^{\rho/\sigma}))$.

We solve the system of equations (17) for the option values (Ξ_{nt}, Ω_{it}) conditional on observed population (R_{nt}) , employment (L_{it}) , commuting costs (K_{int}) , and migration frictions (θ_t) .⁴² We can recover unique (Ξ_{nt}, Ω_{it}) that rationalize the observed changes in the mass of workers without using any information on the unobserved characteristics and without making assumptions about the strength of agglomeration forces.

5.3 Step #3: Estimation of Agglomeration Parameters (α , β)

Next, we back out fundamental productivity (a_{it}) and amenities (b_{nt}) by using observed employment and population density, according to the inverted option values (Ξ_{nt}, Ω_{it}) . Given the agglomeration forces (α, β) , we use (15) and (16) to derive the fundamentals by location for each period $t = 1955, 1960, \cdots, 1975$. Then, we assume that fundamental productivity and amenities consist of location-fixed components $(\{a_i^F\}, \{b_n^F\})$, time-trend components $(\{a_t^*\}, \{b_t^*\})$, and time-varying errors $(\{a_{it}^{Var}\}, \{b_{nt}^{Var}\})$:

$$\ln a_{it} = \ln a_i^F + \ln a_t^* + \ln a_{it}^{\text{Var}}, \quad \ln b_{nt} = \ln b_n^F + \ln b_t^* + \ln b_{nt}^{\text{Var}}.$$
 (18)

The location-specific productivity and amenities capture the fundamental advantages of locations and the trends of productivity and amenities reflect the change in their levels over time within the city. $(\{a_{it}^{Var}\}, \{b_{nt}^{Var}\})$ are the structural residuals in our model that allow us to perfectly match the observed population and employment distributions.

Averaging out the trend terms and taking differences between two consecutive periods, we have:

$$\Delta \ln \left(\frac{a_{it}}{\widetilde{a}_t}\right) = \Delta \ln \left(\frac{a_{it}^{\text{Var}}}{\widetilde{a}_t^{\text{Var}}}\right), \quad \Delta \ln \left(\frac{b_{nt}}{\widetilde{b}_t}\right) = \Delta \ln \left(\frac{b_{nt}}{\widetilde{b}_t^{\text{Var}}}\right), \quad (19)$$

where we denote the geometric mean across locations as $\tilde{a}_t = \exp\left(\frac{1}{N}\sum_{i\in\mathcal{C}}\ln a_{it}\right)$.

The structural residuals of productivity and amenities in (19) difference out both common trends across all blocks in the city in each year and time-invariant locational advantages. Using

⁴²The solution is up to scale because equations (17) exploit only information on the relative migration probabilities across blocks within the city. Since we take the total population of Hiroshima City from the data and assume that the outside utility (u_{ot}) adjusts to rationalize it (see Subsection 5.1), we do not need to determine the absolute levels of $\{\Xi_{nt}\}$ and $\{\Omega_{it}\}$ governing migration between Hiroshima City and the outside world. We normalize the geometric mean of $\{\Xi_{nt}\}$ and $\{\Omega_{it}\}$ to one.

(19), we consider the following moment conditions:

$$\mathbb{E}[\Delta \ln(a_{it}/\widetilde{a}_t) \times \mathbb{1}_i(k)] = 0,$$

$$\mathbb{E}[\Delta \ln(b_{nt}/\widetilde{b}_t) \times \mathbb{1}_n(k)] = 0,$$
(20)

where $\mathbb{1}_n(k)$ is an indicator such that location *n* is in grid *k*, where the grid is defined based on the distance from the CBD. We define five grid cells based on the distance from the CBD and equally allocate blocks into these grid cells in our baseline specification.⁴³ We use the moment conditions (20) to estimate the parameters for agglomeration forces.

Our identification assumption for using the moment conditions (20) is that log changes in the idiosyncratic fundamental productivity and amenities terms are not correlated with distance from the city center. In other words, the systematic change in the gradient of economic activity relative to the distance from the CBD is explained, on average, by the mechanisms of the model rather than by systematic changes in the pattern of structural residuals (19). This identification assumption seems plausible in post-recovery Hiroshima because the spatial extent of our study is small and all blocks in the data experienced similar changes in the economic and political environment.⁴⁴

We assess the validity of our moment conditions in the following two ways. First, to confirm that changes in fundamental amenities and productivity are not correlated with distance from the city center, we plot them against the distance from the city center. Figure D.2 plots the changes during our estimation sample period (1955–1975) and Figure D.3 plots the changes for the period 1950–1955, which are not used for our estimation. In both figures, the moment conditions appear plausible as the changes are generally independent of the distance from the CBD. Second, since the fundamental amenities and productivity are likely to be more homogeneous within a small geographic area, we also estimate the set of parameters using only blocks within 3 kilometers of the CBD. In particular, this addresses the potential concern of post-recovery suburbanization driven by systematic increases in attractiveness for either production or residence further away from the CBD.⁴⁵

Table 2 reports the estimation results using the two-step generalized method of moments (GMM). Columns (1) and (2) report our baseline estimates of the agglomeration parameters for productivity (α) and amenities (β), respectively. Overall productivity (A_{it}) in the workplace rises by around 19 percent when current employment density doubles. Turning to amenities, doubling population density is associated with an 18 percent increase in the value of amenities. In Columns

 $^{^{43}}$ We carry out a robustness check for the sensitivity of estimates to the number of grid cells (using ten cells). See Table D.2 in Appendix D.5.

⁴⁴The effect of radioactivity faded away quickly (see Section 2.1), and hence is unimportant for changes in amenities and productivity, as our estimation data start in 1955.

⁴⁵We also show that fundamental productivity and amenities were not dramatically changed by the atomic bombing by comparing the 1930s and 1975. We assume that the population distribution in 1930 is in a steady state and fundamental productivity and amenities in the 1930s are estimated to rationalize the population distribution in 1936 and employment distribution in 1938. See Figure D.4 for the results.

	(1)	(2)	(3)	(4)
	Productivity	Amenities	Productivity	Amenities
Elasticity of employment density (α)	0.193^a (0.0001)		0.196^a (0.0002)	
Elasticity of population density (β)		0.184^a (0.0009)		0.203 ^{<i>a</i>} (0.0007)
Sample of blocks	All blocks in the city		Blocks within 3 km of CBD	
Sample of periods	Every 5 years from 1955 to 1975		Every 5 years from 1955 to 1975	
Instruments	5 grids for CBD distance		5 grids for CBD distance	

Table 2:	Generalized	method o	f moments	estimates	for aggl	omeration	parameters
					00		1

Note: This table reports two-step generalized method of moments (GMM) estimates exploiting the moment conditions (20). The Eicker-Huber-White heteroskedasticity-robust standard errors are in parentheses. We use data for five periods (1955, 60, 65, 70 and 75). We define five grid cells according to the distance to the CBD for the moment conditions. In Columns (1) and (2) we use all 174 blocks in the city. In Columns (3) and (4) we use 158 blocks that lie within 3 kilometers of the CBD. ^{*a*} indicates significance at the 1 percent level.

(3) and (4) in Table 2 we show similar results when we restrict our sample to blocks within 3 kilometers of the CBD.

Since we estimate strong positive agglomeration forces in both amenities and productivity, the model could have multiple equilibria in light of Proposition 1. Although the direct comparison is difficult due to differences in models and empirical contexts, our estimates of agglomeration forces are broadly in line with those in the existing literature. Our estimated elasticity of productivity with respect to employment density is 0.19. While larger than the 0.03–0.08 from the survey by Rosenthal and Strange (2004), this is relatively close to several recent estimates (e.g., Kline and Moretti 2014; Heblich, Redding, and Sturm 2020; Tsivanidis 2022; Allen and Donaldson 2022) and also well within the range of estimates in the meta-analysis by Melo, Graham, and Noland (2009). Our estimated elasticity of amenities with respect to population density is 0.18. This value is close to the estimates of Ahlfeldt et al. (2015) and Heblich, Redding, and Sturm (2020), while smaller than Tsivanidis (2022).

5.4 Robustness of Agglomeration Parameter Estimates

Instruments based on Pre-war Population Density A potential concern with defining instruments based on distance to the CBD is that the results could be sensitive to the definition of the city center. To address this, we instead use population density in 1936 to define the grid cells. We report the estimation results in Table D.2 in Appendix D.5. We find similar results: the agglomeration parameters for productivity and amenities are 0.178 and 0.165, respectively.

Spatial Spillovers in Productivity and Amenities So far, we have assumed that agglomeration forces in productivity and amenities are at work only in the local block. While this is consistent with empirical evidence that agglomeration forces are highly localized (e.g., Arzaghi and Hender-

son 2008; Ahlfeldt et al. 2015; Gechter and Tsivanidis 2023), productivity in each block may also depend on employment of surrounding blocks. To consider the spatial spread of spillovers, we estimate the agglomeration forces when productivity and amenities are a function of employment and population density, with weights decreasing exponentially with travel time. Specifically, following Ahlfeldt et al. (2015), productivity in block *i* is: $A_{it} = a_{it} \left[\sum_{i' \in C} e^{-\delta \tau_{ii'}} \left(\frac{L_{i't}}{S_{i'}} \right) \right]^{\alpha}$, where δ is a parameter characterizing the spatial decay of productivity and τ_{in} is travel time between blocks. When $\delta \to \infty$, there is no spatial spread of spillovers as in our baseline specification. We specify amenities in an analogous way. Figure D.5 in Appendix D.5 shows the estimated values of the agglomeration parameters (α , β) given different values of spatial decay (δ). As we can see in the figure, the estimated values of the agglomeration parameters range from 0.18 to 0.26 for productivity and from 0.17 to 0.22 for amenities, which are close to the baseline estimates.

Lagged Effects of Agglomeration Forces Our main model assumes that the amenities and productivity of each block depend on its current population and employment densities. However, they could also depend on its past population or employment densities. First, current productivity could reflect the histories of capital, public goods and innovation as determined by past economic activities; second, current amenities could depend on the stock of housing or local infrastructure that is related to past population.⁴⁶ In addition to migration frictions, these effects may also induce history dependence. To take this into account, we specify productivity and amenities in period t as a function of current employment and population densities and the previous employment and population densities in period t - 1, following Allen and Donaldson (2022). We estimate the parameters characterizing both the current and lagged spillover effects using similar moment conditions. Table D.3 in Appendix D.5 shows the results. The elasticity of productivity with respect to current employment density is 0.228 and historical employment density is -0.064. The elasticity of amenities with respect to current population density is 0.175 and historical population density is 0.015. Overall, the influence of lagged population and employment density is small relative to that of current density, and the strength of contemporaneous density remains similar to our baseline estimates.⁴⁷

 $^{^{46}}$ Allen and Donaldson (2022) provide microfoundations for these specifications. In a previous version of this paper, we also provided alternative microfoundations in the within-city context.

⁴⁷Our estimates of agglomeration forces in productivity are broadly similar to those in Allen and Donaldson (2022), which uses long-run county-level data from the U.S. Yet, they find negative contemporaneous agglomeration forces in amenities. This difference may arise from the difference in spatial extent. The negative agglomeration forces in amenities may capture congestion in a local housing market, while our estimates of positive agglomeration forces may capture consumption externalities or neighborhood network effects.

5.5 Accounting for the Recovery: Location Choice for 1950

We are now in position to assess how well our calibrated model fits population and employment changes during the recovery period from 1945 to 1950, data which were not used for calibration. To this end, we evaluate how well the endogenous component of location advantages and the time-invariant unobserved characteristics can fit the workplace and residence choices of 1950, the first location choices made in our model after the atomic bombing. Intuitively, we evaluate how much of the incentive to work or live in a given location during the recovery period can be explained by our model. See Appendix D.6 for more details.

We first use equations (15) and (16) to construct the predicted option values of each location as a residence $(\Xi_{n,1950}^{F})$ and workplace $(\Omega_{i,1950}^{F})$, then substitute them into equation (17) to solve for predicted population and employment in 1950. By construction, the option values in our model are a composite of (i) location-fixed advantages, (ii) the endogenous components of agglomeration forces, (iii) future option values associated with the location, and (iv) idiosyncratic shocks. Among these, factors (i) – (iii) capture the location advantages that our model can explain. In contrast, the idiosyncratic terms (iv), corresponding to ($\{a_{it}^{Var}\}, \{b_{nt}^{Var}\}$) in equation (19), are structural residuals required perfectly match the model prediction with the observed data, which absorb any other characteristics unrelated to the model specification.

Therefore, to evaluate how well our model can fit the recovery, we exclude the structural errors when constructing the model-predicted option values for residence $\{\Xi_{n,1950}^F\}$ and work-place choices $\{\Omega_{i,1950}^F\}$. In obtaining the predicted location decisions for the recovery period (1945–1950), we use the parameter values from our main calibration except we set a higher migration opportunity $\theta_{1950} = 0.9$ as available evidence suggests the mobility rate was substantially higher, possibly for war-related reasons such as job loss, housing destruction, or the end of temporal reallocation during the war (see Appendix A for more details). Importantly, we assume the block-fixed amenities and productivity, (a_{i1950}, b_{n1950}) , equal the averages estimated for the post-recovery period 1955–1975.⁴⁸

Figure 5 illustrates the population and employment distributions predicted by our model for 1950. The horizontal axis is distance to the CBD, and the vertical axis shows population and employment density after the bombing in 1945, as observed in 1950 and as predicted by the model for 1950. For both population and employment, we find that our calibrated model successfully predicts the recovery of the city center, which we indeed observe in the data. The linear regression of the log of observed population (employment) density in 1950 on predicted population (employment) density by the model yields a coefficient of 0.88 (1.01) with a high R-squared around

⁴⁸The validity of this assumption can be checked by comparing the 1955–1975 fixed amenities and productivity with the 1950 exogenous amenities and productivity, which we can compute by netting out the agglomeration forces and future values from the option values ($\{\Xi_{n,1950}^F\}$, $\{\Omega_{i,1950}^F\}$). We find they are quite similar, with correlation coefficients of 0.98 for productivity and 0.99 for amenities.



Figure 5: Recovery of population and employment: Endogenous part explained by our model

Note: Each figure overlays observed log population density (Panel a) and employment density (Panel b) with local polynomial regressions using each on distance from the CBD. We estimate three separate regressions: the 1945 population and employment densities (small dashed line); the observed 1950 population and employment densities (long dashed line); and the 1950 population and employment densities inferred under the counterfactual scenario in which we exclude the structural error components of amenities and productivity (solid line). Each dot represents a block, with different colors for the predicted density and the observed density.

 $0.88~(0.91).^{49}$ This result shows that our calibrated model successfully explains the fast and strong recovery from 1945 to 1950. 50

6 The Role of Agglomeration Forces

Having demonstrated that our calibrated model can account for the recovery of central Hiroshima, we now analyze the role of agglomeration forces in the recovery. In Subsection 6.1 we undertake a counterfactual experiment in which we exclude agglomeration forces in both productivity and amenities from our calibrated model. In Subsection 6.2 we investigate the existence of multiple equilibria. Consistent with the importance of agglomeration forces, we numerically find an alternative equilibrium in which the city center did not recover. This suggests that expectations can be self-fulfilling by selecting the recovery equilibrium among multiple equilibria.

⁴⁹Note that the prediction of our model is substantially more accurate than just capturing the general tendency that blocks closer to the city center tend to have higher density in 1950. Indeed, regressing log population (employment) density on the log distance from the city center yields R-squared around 0.20 (0.49), which is considerably lower than the R-squared from our model prediction.

⁵⁰Note that we exclude idiosyncratic components of locational fundamentals that may capture factors such as the recovery plan or property rights. Yet, these shocks may not be essential in inducing the recovery (see also Section 7).

6.1 Agglomeration Forces as the Key Driver of the Recovery

The city center recovers when individuals regard it as an attractive residence and workplace. Strong agglomeration forces can be a primary source of attractiveness. These forces operate by increasing the expected population and employment density, which in turn leads to improved amenities and productivity. An alternative possibility is that the city center has attractive location fundamentals so that it attracts population and employment regardless of agglomeration forces. Which forces induced the recovery of central Hiroshima in our calibrated model? To investigate this, we compute counterfactual population and employment distributions for 1950 when spillovers in amenities and productivity are absent.

We solve the model for counterfactual equilibrium 1950 population and employment distributions, using the same parameter values as Subsection 5.5 but setting both agglomeration parameters, α and β , to zero.⁵¹ As in the baseline, individuals make forward-looking migration decisions taking into account future fundamental productivity, amenities, and commuting costs. Notably, as in Subsection 5.5, we assume that the fundamental amenities and productivity during the recovery period equal the average amenities and productivity during 1955-1975.⁵² If agglomeration forces play the key role in explaining the attractiveness of the city center in our model, then this counterfactual exercise would not be able to predict the recovery.⁵³

Figure 6 shows the counterfactual population and employment densities in the absence of agglomeration forces. The model no longer predicts the recovery of population and employment in central Hiroshima. This is in stark contrast to our main calibrated model in Figure 5. Given that the only deviation from our main calibrated model is the shutdown of agglomeration forces, this result highlights that agglomeration forces play the key model role in explaining the recovery of the city center. Note that as discussed in the last paragraph of Section 3, the importance of agglomeration forces is in line with our reduced-form results that the observed fundamental location characteristics, which are independent of agglomeration forces, do not explain the recovery.

⁵¹To focus on the population and employment distributions within the city, we assume in this counterfactual that the total population matches the observed data.

⁵²This assumption on fundamental amenities and productivity is important because even without agglomeration forces our model can fit any population and employment distribution in 1950 as long as the structural errors (a_{i1950}, b_{n1950}) can take any value. Therefore, the role of agglomeration forces highlighted in this subsection presumes that the levels of fundamental amenities and productivity are similar to those in the post-recovery period 1955–1975. Consistent with this assumption, we find that the values of the structural errors (a_{i1950}, b_{n1950}) are similar to the 1955–1975 fundamentals. See previous discussion in Subsection 5.5.

⁵³The parameters (α , β) capture not only pure externalities of density but also other channels through which population or employment density affects productivity and amenities. In this counterfactual, we turn off all these density effects simultaneously.

Figure 6: Population and employment distributions with no agglomeration forces



Note: Each figure overlays log population density (Panel a) and employment density (Panel b) with local polynomial regressions of each on distance from the CBD. We run three separate regressions: one for the observed 1945 population and employment densities (small dashed line), one for the observed 1950 population and employment densities (long dashed line), and one for the inferred 1950 population and employment densities when we shut down agglomeration forces in both productivity and amenities (solid line). Each dot represents a block, with different colors for the predicted density and the observed density.

6.2 Multiple Equilibria and Self-fulfilling Expectations of Recovery

This section examines an alternative equilibrium in which the recovery does not occur. When agglomeration forces are important, the model may have multiple equilibria because whether the city center remains attractive depends on whether people expect a high density city center in the near future. There could exist an alternative equilibrium in which central Hiroshima does not recover. If so, the selection of the recovery equilibrium among multiple equilibria might be crucial in explaining the recovery of central Hiroshima.

Specifically, we present an example of an alternative rational-expectations equilibrium in which the city center does not recover. We use the parameters values and fundamentals estimated in Section 5 and assume that the total population matches the observed data. To find an alternative equilibrium, we start off with guesses of population and employment in 1975 that are different from the observed data.⁵⁴ We then solve for the dynamics of population and employment consistent with equilibrium conditions dating back to 1945. Subsequently, we update our initial guess for 1975 until the backward solution in 1945 converges to the initial conditions observed in 1945.

Figure 7 provides a visualization of population and employment densities in an alternative

⁵⁴To construct initial guesses for the population and employment distributions in 1975, we simulate population and employment dynamics forward in time from 1950, assuming that people have a myopic expectation that the population and employment distribution in period t + 1 will be the same as in period t.

equilibrium. We compare these with the realizations of population and employment in 1950 and the initial pattern of 1945. We label a block with a high concentration of population and employment in this counterfactual as an alternative CBD.⁵⁵ In Panel (a) we find that pre-war central Hiroshima does not recover, and its population density is even lower than the initial level in 1945. In Panel (b) we find a similar pattern for employment. These results are consistent with the idea that a totally different city structure could have emerged as an alternative equilibrium.



Figure 7: Population and employment distribution in an alternative equilibrium

Note: Each figure plots log population density (Panel a) and employment density (Panel b) with local polynomial regressions of each on distance from the CBD. We run three separate regressions: one for the observed 1945 population and employment densities (small dashed line), one for the observed 1950 population and employment densities (long dashed line), and one for the inferred 1950 population and employment densities in an alternative equilibrium (solid line) when people expect that the pre-war CBD will not recover and an alternative block located at the vertical dashed line will grow. Each dot represents a block, with different colors for the predicted density and the observed density. The location with growing population and employment density is labeled "alternative CBD".

This multiplicity of equilibria highlights the potential importance of self-fulfilling expectations as an equilibrium selection device (Krugman 1991; Matsuyama 1991). In our rational-expectations model, if people expect the recovery, then the recovery equilibrium realizes because such expectations make the city center an attractive residence and workplace due to agglomeration forces. In contrast, if people do not expect the recovery of the city center, then no recovery equilibrium realizes as the city center remains unattractive. Therefore, our result highlights that the formation of expectations in recovery might be crucial in inducing the recovery of central Hiroshima after the bombing.

⁵⁵This block (*Niho machi*) is a plausible candidate for an alternative CBD as it hosts high productivity firms such as *Toyo Kogyo* (currently known as a large automotive manufacturer, Mazda Motor) and is close to Hiroshima port.

7 Discussion on the Origin of Expectations in the Recovery

Our analysis has demonstrated that the expectation that blocks with high pre-war density would regain high density post-war is crucial in explaining the recovery of central Hiroshima. We now discuss the potential factors contributing to the formation of expectations, while we remain agnostic about *why* the expectations of recovery emerged.

Government Recovery Plan First, the presence of a government recovery plan would have facilitated the formation of expectations, despite the fact that publication of the plan lagged the onset of the recovery.⁵⁶ Moreover, the government was substantially underfunded, as discussed in Subsection 3.3, and implementation of the plan faced substantial difficulty.⁵⁷

Land Ownership Land ownership is an additional factor to consider. Yet, the recovery is unlikely to be explained by a strong tendency of original landowners to return to their prior homes. In particular, personal land ownership was quite rare in pre-WWII Japan; the rate of land ownership in pre-WWII urban area in Japan was likely less than 10 percent (Kato 1988).⁵⁸ In our context, unlike conventional air raids, the death rate near the epicenter of the atomic bombing was nearly 100 percent, implying that the number of surviving landowners was small.⁵⁹ To further analyze the influence of land ownership on the recovery quantitatively, we carry out a counterfactual analysis in which we assume that landowners consisted of 20 percent of the total population in 1936 and those who survived the bombing returned to their pre-bombing block by 1950 due to personal attachment. Without agglomeration forces, however, we do not find a strong recovery of central Hiroshima relative to the observations (see details in Appendix D.6). That said, even the small number of surviving landowners that returned to their original locations may have played an important role in forming expectations in the recovery.

Salient Location Characteristics Another possibility is that the tangible presence of some location characteristics in the city center may have induced an anchoring of expectations. The first example is the transportation system, especially the tram network. While the direct benefit of access to trains does not appear to be essential to the recovery as we control for transportation

⁵⁶It is unlikely that zoning laws contributed to the recovery in 1945–50 because the first post-war zoning of Hiroshima was published in 1949, when the recovery was nearly complete. In addition, there were only four types of zones (commercial, residential, industrial, and unspecified), and the zoning allowed substantial mixed land use (Asano 2012).

⁵⁷Appendix B.2 reports that controlling for the number of public housing units supplied during 1945–1950 does not alter our regression results, which would be plausible given the limited supply due to the budget shortage.

⁵⁸The fraction of households owning a home was also quite low in pre-WWII urban Japan at around 25 percent (Hinokidani and Sumita 1988).

⁵⁹The turnover of business owners was also high. According to Hiroshima City Government (1983), in 1958, approximately 28 percent of stores on a notable shopping street, *Hondori*, remained in the same location as before the war, while the remaining 72 percent of stores had begun operating only after the bombing.

access in both the reduced-form and structural analyses, the relatively quick restoration of the pre-war train network may have anchored people's expectations of reconstruction. Another example is Hiroshima castle. Although the castle itself was completely destroyed by the bombing and was unlikely to provide direct amenity value, its historical salience may have made it difficult for people to expect a situation in which the city center moves away from the castle.

Narratives Lastly, the narrative of "rebuilding from the atomic bombing" may have sounded like a compelling success story and been shared widely (Shiller 2017).⁶⁰ As long as individuals were aware that many others shared this narrative, they could expect that the city structure would look like the pre-war Hiroshima in their memory, thereby inducing the recovery of the pre-war city center.⁶¹

Summary These underlying factors may influence the attractiveness of each location by either directly improving location-specific amenities (b_{nt}) and productivity (a_{it}) , or via expectation channels. However, as discussed in Subsections 5.5 and 6.1, the ability of our calibrated model to explain the recovery comes from agglomeration forces, not location-specific amenities and productivity. In the expectations channel, several factors may alter expectations regarding the future population and employment distributions after the bombing, thereby affecting the attractiveness of each location via agglomeration forces. The self-fulfilling nature of expectations found in Subsection 6.2 suggests the possibility that the above-mentioned factors may have played a key role in inducing the recovery through the expectations channel.

8 Conclusion

A major source of public policy debate is the resilience of cities in the face of large shocks. To shed light on this question, we examine the atomic bombing of Hiroshima, which drastically changed the city's structure by completely destroying the city center while sparing the city's outskirts. We collect and digitize new historical data on Hiroshima's population, employment, wartime destruction, and fundamental location characteristics at the city block level. Then, we document the strong resilience of Hiroshima's city structure: the destroyed city center recovered its population density just five years after the atomic bombing. Our reduced-form analysis reveals that controlling for prominent observable location characteristics, such as altitude, access to natural water

⁶⁰Although it is challenging to empirically assess how powerful and widespread such a narrative was, the 1946 Statistical Abstract of Hiroshima is suggestive in stating '… rumors like "nothing will grow here for 75 years" immediately disappeared among people with their burning desire to rebuild…' (p. 4, translated by the authors).

⁶¹This relates to the idea of "memory-based expectations," in which people form expectations based on their past experiences (Malmendier and Wachter 2022). To gauge its potential importance, we simulate the model as in Subsection 5.5 assuming that workers expect the population distribution in 1936 and employment distribution in 1938 to be realized again in 1950. The simulation shows that such purely memory-based expectations also induce the recovery of the city center, suggesting that such expectations are comparable to rational expectations in our context.

and train stations, does not explain the recovery.

To identify the mechanism behind the recovery, we develop and calibrate a novel dynamic quantitative model of internal city structure that incorporates commuting, forward-looking migration decisions, migration frictions, agglomeration forces, and heterogeneous fundamentals across locations. Estimating the model with post-recovery data (1955–1975), we find strong agglomeration forces in productivity and amenities. The calibrated model successfully explains the resurgence of population and employment in the city center after the bombing (1945–1950), and agglomeration forces are essential for this success.

In the presence of strong agglomeration forces, multiple equilibria may exist because the city center does not become attractive if it is not expected to achieve high density. We find that there exists an alternative forward-looking equilibrium where the city center fails to recover. This suggests that the recovery crucially depended on people's expectations, as they can be self-fulfilling and select the equilibrium of recovery. We argue that certain factors, such as government recovery plans, the anchoring effect of salient location characteristics in the city center, property rights, and popular narratives of rebuilding, may have led people to expect that the destroyed areas would once again achieve high density as in the pre-war period. Taken together, our quantitative findings highlight the role of agglomeration forces, multiple equilibria, and expectations in the resilience of city structure.

Beyond the context of Hiroshima's resilience, agglomeration economies and expectations are important determinants of the dynamics of city structure. Therefore, our findings suggest that policymakers may be able to substantially change the dynamics of city structure if they could influence expectations about future city structure. Our theoretical framework developed in this paper could serve as a useful starting point for performing quantitative analyses to understand how the organization of economic activities within cities evolves over time. However, our model does not incorporate an explicit process in which people form expectations. Developing additional understanding of the ways in which agents can form expectations about a city's future structure is an interesting area for further research.

References

- Ahlfeldt, Gabriel M, Stephen J Redding, Daniel M Sturm, and Nikolaus Wolf. 2015. "The economics of density: Evidence from the Berlin Wall." *Econometrica* 83 (6):2127–2189.
- Allen, Treb and Dave Donaldson. 2022. "Persistence and Path Dependence in the Spatial Economy." *NBER Working Paper* No. 28059.
- Almagro, Milena and Tomás Domínguez-Iino. 2022. "Location sorting and endogenous amenities: Evidence from Amsterdam." *Unpublished manuscript* .
- Ambrus, Attila, Erica Field, and Robert Gonzalez. 2020. "Loss in the time of cholera: long-run

impact of a disease epidemic on the urban landscape." *American Economic Review* 110 (2):475–525.

- Arzaghi, Mohammad and J Vernon Henderson. 2008. "Networking off Madison Avenue." *Review* of Economic Studies 75 (4):1011–1038.
- Asano, Jun-ichiro. 2012. "A Study on Method of the First Zoning Designation by the City Planning Law and the Building Regulation Law of 1919: In the Cases of 43 Small and Middle Scale Cities Where the City Planning Law was Enforced by the Taisyo Era." *Journal of Architecture Planning* 77:2389–2398. (In Japanese).
- Balboni, Clare. 2021. "In harm's way? infrastructure investments and the persistence of coastal cities." *Unpublished manuscript*.
- Baldwin, Richard E. 2001. "Core-periphery model with forward-looking expectations." *Regional Science and Urban Economics* 31 (1):21–49.
- Barreda-Tarrazona, Iván, Tapas Kundu, and Stein Østbye. 2021. "On rational forward-looking behavior in economic geography: An experimental analysis." *Regional Science and Urban Economics* 87:103654.
- Bleakley, Hoyt and Jeffrey Lin. 2012. "Portage and path dependence." Quarterly Journal of Economics 127 (2):587-644.
- Bosker, Maarten, Steven Brakman, Harry Garretsen, and Marc Schramm. 2007. "Looking for multiple equilibria when geography matters: German city growth and the WWII shock." *Journal of Urban Economics* 61 (1):152–169.
- Brakman, Steven, Harry Garretsen, and Marc Schramm. 2004. "The strategic bombing of German cities during World War II and its impact on city growth." *Journal of Economic Geography* 4 (2):201–218.
- Brooks, Leah and Byron Lutz. 2019. "Vestiges of transit: Urban persistence at a microscale." *Review* of *Economics and Statistics* 101 (3):385–399.
- Brooks, Leah, Jonathan Rose, and Stan Veuger. 2022. "Destruction, Policy, and the Evolving Consequences of Washington, DC's 1968 Civil Disturbance." *FRB of Chicago Working Paper*.
- Caliendo, Lorenzo, Maximiliano Dvorkin, and Fernando Parro. 2019. "Trade and labor market dynamics: General equilibrium analysis of the China trade shock." *Econometrica* 87 (3):741–835.
- Ciccone, Antonio and Robert Hall. 1996. "Productivity and the Density of Economic Activity." *American Economic Review* 86 (1):54–70.
- Conley, Timothy G. 1999. "GMM estimation with cross sectional dependence." Journal of Econometrics 92 (1):1-45.
- Davis, Donald R and David E Weinstein. 2002. "Bones, bombs, and break points: the geography of economic activity." *American Economic Review* 92 (5):1269–1289.
- ----. 2008. "A search for multiple equilibria in urban industrial structure." Journal of Regional Science 48 (1):29-65.
- Dingel, Jonathan I and Felix Tintelnot. 2020. "Spatial economics for granular settings." Tech. rep., National Bureau of Economic Research.

- Duranton, Gilles and Diego Puga. 2004. "Micro-foundations of urban agglomeration economies." In Handbook of regional and urban economics, vol. 4. Elsevier, 2063–2117.
- Feigenbaum, James, James Lee, and Filippo Mezzanotti. 2022. "Capital destruction and economic growth: The effects of Sherman's March, 1850–1920." American Economic Journal: Applied Economics 14 (4):301–42.
- Fujita, Masahisa, Paul R Krugman, and Anthony Venables. 1999. The spatial economy: Cities, regions, and international trade. MIT press.
- Fujita, Masahisa and Hideaki Ogawa. 1982. "Multiple equilibria and structural transition of nonmonocentric urban configurations." *Regional Science and Urban Economics* 12 (2):161–196.
- Fujita, Masahisa and Jacques-François Thisse. 1996. "Economics of agglomeration." *Journal of the Japanese and International Economies* 10 (4):339–378.
- Fukao, Kyoji and Roland Bénabou. 1993. "History versus expectations: a comment." *Quarterly Journal of Economics* 108 (2):535–542.
- Garcia-Cicco, Javier, Roberto Pancrazi, and Martin Uribe. 2010. "Real business cycles in emerging countries?" *American Economic Review* 100 (5):2510–2531.
- Gechter, Michael and Nick Tsivanidis. 2023. "Spatial spillovers from high-rise developments: Evidence from the Mumbai Mills." *Unpublished manuscript*.
- Glaeser, Edward L. 2022. "Urban resilience." Urban Studies 59 (1):3-35.
- Halleck Vega, Solmaria and J Paul Elhorst. 2015. "The SLX model." *Journal of Regional Science* 55 (3):339–363.
- Harada, Masataka, Gaku Ito, and Daniel M Smith. forthcoming. "Destruction from above: Long-term legacies of the Tokyo air raids." *Journal of Politics*.
- Hayakawa, Kazuo and Michiko Nakaouji. 1965. "Toshi no jinko kibo to chika no kankei ni tsuite (On the relationship between population size and land prices)." *Nihon kenchiku gakkai ronbun houkoku shu* Extra volume. (In Japanese).
- Heblich, Stephan, Stephen J Redding, and Daniel M Sturm. 2020. "The making of the modern metropolis: evidence from London." *Quarterly Journal of Economics* 135 (4):2059–2133.
- Heblich, Stephan, Alex Trew, and Yanos Zylberberg. 2021. "East-Side Story: Historical Pollution and Persistent Neighborhood Sorting." *Journal of Political Economy* 129 (5):1508–1552.
- Hinokidani, Mieko and Shyoji Sumita. 1988. "Jutaku Syoyu Keitai No Henyo Katei Ni Kansuru Kenkyu (A Study on the Changing Process of Housing Tenure)." *Journal of Architecture and Planning* 392:136–146. (In Japanese. The English title and abstract are available.).
- Hiroshima City Government. 1971. "Hiroshima Genbaku Sensai-Shi (The Hiroshima Atomic Bombing Damage Records)." (In Japanese).
- ----. 1983. "Hiroshima Shin-Shi: Chiri-hen (The New History of Hiroshima: The Geography)." (In Japanese).
- Holmes, Thomas J. 1999. "How industries migrate when agglomeration economies are important." *Journal of Urban Economics* 45 (2):240–263.

- Hornbeck, Richard and Daniel Keniston. 2017. "Creative destruction: Barriers to urban growth and the Great Boston Fire of 1872." *American Economic Review* 107 (6):1365–98.
- Kato, Yuriko. 1988. "On the ownership of land and housing in pre-war Tokyo (Senzen no toukyou ni okeru tochi kaoku no shoyuu jyoukyou ni tsuite)." *Journal of Aoyama Gakuin Woman's Junior College* 42:99–114. URL https://cir.nii.ac.jp/crid/1390009224575908352. (In Japanese).
- Kleinman, Benny, Ernest Liu, and Stephen J Redding. 2023. "Dynamic spatial general equilibrium." *Econometrica* 91 (2):385–424.
- Kline, Patrick and Enrico Moretti. 2014. "Local economic development, agglomeration economies, and the big push: 100 years of evidence from the Tennessee Valley Authority." *Quarterly Journal of Economics* 129 (1):275–331.
- Krugman, Paul. 1991. "History versus expectations." Quarterly Journal of Economics 106 (2):651–667.
- ----. 1993. "First nature, second nature, and metropolitan location." *Journal of Regional Science* 33 (2):129-144.
- Lin, Jeffrey and Ferdinand Rauch. 2022. "What Future for History Dependence in Spatial Economics?" *Regional Science and Urban Economics* 94:103628.
- Lucas, Robert E and Esteban Rossi-Hansberg. 2002. "On the internal structure of cities." *Econometrica* 70 (4):1445–1476.
- Malmendier, Ulrike and Jessica A Wachter. 2022. "Memory of Past Experiences and Economic Decisions." *Unpublished manuscript*.
- Matsuyama, Kiminori. 1991. "Increasing returns, industrialization, and indeterminacy of equilibrium." *Quarterly Journal of Economics* 106 (2):617–650.
- Melo, Patricia C, Daniel J Graham, and Robert B Noland. 2009. "A meta-analysis of estimates of urban agglomeration economies." *Regional Science and Urban Economics* 39 (3):332–342.
- Michaels, Guy and Ferdinand Rauch. 2018. "Resetting the urban network: 117–2012." *The Economic Journal* 128 (608):378–412.
- Miguel, Edward and Gerard Roland. 2011. "The long-run impact of bombing Vietnam." *Journal of Development Economics* 96 (1):1–15.
- Monte, Ferdinando, Charly Porcher, and Esteban Rossi-Hansberg. 2023. "Remote Work and City Structure." .
- Monte, Ferdinando, Stephen J Redding, and Esteban Rossi-Hansberg. 2018. "Commuting, migration, and local employment elasticities." *American Economic Review* 108 (12):3855–90.
- Murphy, Kevin M, Andrei Shleifer, and Robert W Vishny. 1989. "Industrialization and the big push." *Journal of Political Economy* 97 (5):1003–1026.
- Nagasaki City Government. 1977. "Nagasaki Genbaku Sensai-Shi Dai 1-kan Sousetsuhen (The Nagasaki Atomic Bombing Damage Records: Volume 1 Overview)." (In Japanese).
- Nozawa, Hiroshi. 1934. "Hiroshima-shi no Hattatsu to sono Zinbun Gensyo no Chiri-teki Kousatsu (A Geographical Consideration on the Development of Hiroshima city and its Humanilistic Issues)." *Chiri Ronso* 5:47–91. (In Japanese).

- Ottaviano, Gianmarco IP. 2001. "Monopolistic competition, trade, and endogenous spatial fluctuations." *Regional Science and Urban Economics* 31 (1):51–77.
- Oyama, Daisuke. 2009. "History versus expectations in economic geography reconsidered." Journal of Economic Dynamics and Control 33 (2):394–408.
- Rauch, James E. 1993. "Does history matter only when it matters little? The case of city-industry location." *Quarterly Journal of Economics* 108 (3):843–867.
- Redding, Stephen J and Esteban Rossi-Hansberg. 2017. "Quantitative spatial economics." *Annual Review of Economics* 9:21–58.
- Redding, Stephen J and Daniel M Sturm. 2023. "Neighborhood effects: Evidence from war-time destruction in London." *Unpublished manuscript*.
- Redding, Stephen J, Daniel M Sturm, and Nikolaus Wolf. 2011. "History and industry location: Evidence from German airports." *Review of Economics and Statistics* 93 (3):814–831.
- Rosenthal, Stuart S and William C Strange. 2004. "Evidence on the nature and sources of agglomeration economies." In *Handbook of regional and urban economics*, vol. 4. Elsevier, 2119–2171.
- Rossi-Hansberg, Esteban. 2004. "Cities under stress." Journal of Monetary Economics 51 (5):903–927.
- Schumann, Abel. 2014. "Persistence of Population Shocks: Evidence from the Occupation of West Germany after World War II." American Economic Journal: Applied Economics 6 (3):189–205.
- Shiller, Robert J. 2017. "Narrative economics." American Economic Review 107 (4):967-1004.
- Shinoda, Hideaki. 2008. "Heiwa kochiku to shite no Hiroshima no sengo fukkou (Post-war Recovery of Hiroshima for Peace)." *IPSHU Research Report Series* 40:2–24. (In Japanese).
- Siodla, James. 2015. "Razing San Francisco: The 1906 disaster as a natural experiment in urban redevelopment." *Journal of Urban Economics* 89:48–61.
- Takahashi, Hiroko. 2008. "Genbaku touka ippungo: kesareta zanryuu housyasen no eikyou (Oneminute after atomic bombing: The hidden effect of remaining radiation)." Amerika kenkyuu 42:1–19. (In Japanese. English abstract is available.).
- Takano, Keisuke. 2023. "Path-dependence in the location of business agglomeration: The impact of the postwar land requisition in Yokohama." *Unpublished manuscript* .
- Takezaki, Yoshihiko and Ryoji Soda. 2001. *Hiroshima Genbaku Dezitaru Atorasu (The Digital Atlas of Atomic Bombing in Hiroshima)*. Research center for regional geography, Hiroshima University. (In Japanese).
- Tani, Kenji. 2017. "Konjyaku mappu kyuhan chikeizu tairu gazou haishin etsuran sabisu no kaihatsu (Developing the web viewing service of images of the old version topographic map)." *GIS Riron to Ouyou (GIS Theory and Applications)* 25 (1):1–10. (In Japanese).
- Tsivanidis, Nick. 2022. "Evaluating the impact of urban transit infrastructure: Evidence from Bogota's transmilenio." *Unpublished manuscript*.
- Warnes, Pablo Ernesto. 2021. "Transport infrastructure improvements and spatial sorting: Evidence from buenos aires." *Unpublished manuscript* .
- Yamagishi, Atsushi and Yasuhiro Sato. 2023. "Persistent Stigma in Space: 100 Years of Japan's Invisible Race and Their Neighborhoods." Unpublished manuscript .

CENTRE FOR ECONOMIC PERFORMANCE Recent Discussion Papers

1987	Gustave Kenedi	Beyond the enrolment gap: Financial barriers and high-achieving, low-income students' persistence in higher education
1986	Stephen J. Redding Daniel M. Sturm	Neighborhood effects: Evidence from wartime destruction in London
1985	Tom Kirchmaier Ekaterina Oparina	Under pressure: Victim withdrawal and police officer workload
1984	Melanie Arntz Sebastian Findeisen Stephan Maurer Oliver Schlenker	Are we yet sick of new technologies? The unequal health effects of digitalization
1983	Javad Shamsi	Immigration and political realignment
1982	Ekaterina Oparina Christian Krekel Sorawoot Srisuma	Talking therapy: Impacts of a nationwide mental health service in England
1981	Xiao Chen Hanwei Huang Jiandong Ju Ruoyan Sun Jialiang Zhang	Endogenous mobility in pandemics: Theory and evidence from the United States
1980	Alan Manning Graham Mazeine	Should I stay or should I go? Return migration from the United States
1979	Rosa Sanchis-Guarner José Montalbán Felix Weinhardt	Home broadband and human capital formation
1978	Luca Macedoni John Morrow Vladimir Tyazhelnikov	Firms in product space: Adoption, growth and competition

1977	Gonzalo Nunez-Chaim Henry G. Overman Capucine Riom	Does subsidising business advice improve firm performance? Evidence from a large RCT
1976	Julian Alves Bruno Serra Jason Greenberg Yaxin Guo Ravija Harjai John Van Reenen	Labour market power: New evidence on Non- Compete Agreements and the effects of M&A in the UK
1975	Virginia Minni	Global managers, local workers: Wage setting inside a multinational firm
1974	Jonathan Colmer Suvy Qin John Voorheis Reed Walker	The changing nature of pollution, income and environmental inequality in the United States
1973	Italo Colantone Gianmarco I.P. Ottaviano Piero Stanig	The social footprint of globalization: Towards the introduction of strategic industries in quantitative trade models
1972	Viet Nguyen-Tien Robert J.R. Elliott Eric Strobl Chengyu Zhang	Estimating the longevity of electric vehicles: What do 300 million MOT test results tell us?
1971	Michael Amior Jan Stuhler	Immigration, monopsony and the distribution of firm pay
1970	Jonathan Colmer David Lagakos Martin Shu	Is the electricity sector a weak link in development?
1969	Gaia Dossi Marta Morando	Political ideology and innovation

The Centre for Economic Performance Publications Unit Tel: +44 (0)20 7955 7673 Email <u>info@cep.lse.ac.uk</u> Website: <u>http://cep.lse.ac.uk</u> Twitter: @CEP_LSE